

# Road-, Air- and Water-based Future Internet **Experimentation**

Project Acronym:	RAWFIE		
Contract Number:	645220		
Starting date:	Jan 1st 2015	Ending date:	Dec 31st, 2018

Deliverable Number and Title	D3.2 - Specification & Analy	vsis of RAWFIE Com	ponents Requirements (b)
Confidentiality	PU	Deliverable type <sup>1</sup>	R
Deliverable File	RAWFIE_D_3_2_final	Date	2016-03-15
Approval Status <sup>2</sup>	WP leader	Version	1.000
Contact Person	Nikolaos Pringouris	Organization	HAI
Phone	+30 22620 46572	E-Mail	PRIGGOURIS.Nikolaos@haicorp.com

 <sup>&</sup>lt;sup>1</sup> Deliverable type: P(Prototype), R (Report), O (Other)
 <sup>2</sup> Approval Status: WP leader, 1<sup>st</sup> Reviewer, 2<sup>nd</sup> Reviewer, Advisory Board

Name	Company	E-Mail
Nikolaos Pringouris	HAI	PRIGGOURIS.Nikolaos@haicorp.com
Marcel Heckel	Fraunhofer	marcel.heckel@ivi.fraunhofer.de
Anestis Trypitsidis	AvionTek	aerospace@epsilon.gr
Sotiris Glykofrydis	HMOD	sglikofridis@dideap.mil.gr
Kiriakos Georgouleas	HAI	Georgouleas.Kiriakos@haicorp.com
Kakia Panagidi	UOA	kakiap@di.uoa.gr
Blerina Lika	UOA	b.lika@di.uoa.gr
Kostas Kolomvatsos	UOA	kostasks@di.uoa.gr
José Braga	MST	jbraga@oceanscan-mst.com
Giovanni Tusa	IES	g.tusa@iessolutions.eu
Philippe Dallemagne	CSEM	Philippe.Dallemagne@csem.ch

#### **AUTHORS TABLE**

#### **REVIEWERS TABLE**

Name	Company	E-Mail
Philippe Dallemagne	CSEM	Philippe.Dallemagne@csem.ch
Miltiadis Kyriakakos	UOA	miltos@di.uoa.gr

#### DISTRIBUTION

Name / Role	Company	Level of confidentiality <sup>3</sup>	Type of deliverable
All		PU	R

#### **CHANGE HISTORY**

Version Date Reason for Change Pages/Sections	Version Date Reason for Change Pages/Sections	
---	---	--

<sup>&</sup>lt;sup>3</sup> Deliverable Distribution: PU (Public, can be distributed to everyone), CO (Confidential, for use by consortium members only), RE (Restricted, available to a group specified by the Project Advisory Board).



			Affected
0.001	2015-11-17	First Document Issue with Introduction & ToC.	All
0.002	2015-12-03	Finalize ToC and assignment of responsible partners	All
0.003	2016-01-15	New scenario added in chapter 3, Requirements added in various subsections of chapter 4	Chapter 3, Chapter 4
0.004	2016-01-18	Update in requirements sections	4.2, 4.3
0.005	2016-02-08	Update of Booking tools & booking services requirements	4.1.3
0.006	2016-02-14	Update of Testbed Manager requirements	4.2.6
0.007	2016-02-17	Update of scenarios section	3
0.008	2016-02-18	Update of various requirements sections and accept of changes	All
0.009	2016-03-01	Added and updated requirements for System Monitoring Tools and Resource Explorer Tool	4.1
0.010	2016-03-10	Document issued for internal review	4.1, 4.2, 4.3, 4.4, 5, 6
1.000	2016-03-14	Adaptation/modifications based on internal review comments	All



#### Abstract:

This deliverable comprises the  $2^{nd}$  version of the RAWFIE Components requirements. Based on the grounds of the  $1^{st}$  version and by utilizing information regarding the architecture and detailed design elaborated during the  $1^{st}$  iteration of the project it attempts to refine the list of requirements defined in the previous version of the Requirements document.

The initial version of the Requirements Analysis provided a coarse-grained outline of the overall system, the envisaged components and the expected user and system high-level requirements by defining two broad categories. In the present version, a more elaborated requirement analysis look is attempted. Requirements are classified following the RAWFIE component breakdown structure prescribed by the RAWFIE Architecture related documents compiled during the 1<sup>st</sup> iteration cycle. The methodology used is still based on the VOLERE like card template agreed in the 1<sup>st</sup> version.

In keeping with the overall project workflow, the requirements captured and synthesized here will be provided as input to WP4 – Platform Design for the  $2^{nd}$  development cycle.

The use cases defined in the 1<sup>st</sup> version of the Requirements document remain valid while some additional ones have been added.

A traceability matrix is also provided between the requirements defined during the 1<sup>st</sup> and the 2<sup>nd</sup> version of the Requirement analysis

Keywords: requirements, scenario, experiment, constraints standards & regulations, functional & non-functional



# Part II: Table of Contents

Pa	rt II: 7	Fable	e of Contents	5
]	List of	f Fig	ures	7
]	List of	f Tab	les	8
Pa	rt III:	Exec	cutive Summary	9
Pa	rt IV:	Mair	n Section 1	0
1	Intr	oduc	tion1	1
	1.1	Sco	pe of Deliverable 1	1
-	1.2	Abb	previations1	1
2	Met	thodo	blogy1	4
2	2.1	Gen	eral 1	4
	2.2	Def	initions1	8
3	Use	r Sce	enarios1	9
( (	3.1 (ops).	Scer 19	nario 7 – Gathering Information for Naval Search and Rescue (SAR) Operation	15
	3.2	Scer	nario 8 – Mobilize resources and gather sensor data (1st year review scenario) 2	2
4	Sys	tem a	& Component Requirements 2	6
2	4.1	Plat	form Requirements	8
	4.1.	1	General 2	8
	4.1.	2	Web Portal	0
	4.1.	3	Booking Tool	1
	4.1.	4	System Monitoring Tool	8
	4.1.	5	Resource Explorer Tool 4	0
	4.1.	6	Experiment Authoring Tool 4	-2
	4.1.	7	Experiment Monitoring Tool	0
	4.1.	8	UxV Navigation Tool	1
	4.1.	9	Visualisation Tool	3
	4.1.	10	Data Analysis Tool	6
	4.1.	11	Testbeds Directory Service	9



	4.1.12	EDL Compiler and Validator
	4.1.13	Experiment Validation Service
	4.1.14	Users & Rights Service
	4.1.15	Booking Service
	4.1.16	Launching Service
	4.1.17	Visualisation Engine
	4.1.18	Experiment Controller
	4.1.19	Data Analysis Engine
	4.1.20	System Monitoring Service
	4.1.21	Accounting Service
4	.2 Tes	tbed Requirements
	4.2.1	General
	4.2.2	Monitoring Manager
	4.2.3	Network Controller
	4.2.4	Resource Controller
	4.2.5	Testbed Proxy
	4.2.6	Testbed Manager
4	.3 Ux	V Requirements
	4.3.1	General 107
	4.3.2	UxV Node
	4.3.3	UxV Network and Communication
	4.3.4	UxV Sensor and Localisation
	4.3.5	UxV On-board storage
	4.3.6	UxV On-board processing
	4.3.7	UxV Management
4	.4 Eth	ics and Security Requirements
5	Traceab	ility Mapping 127
6	Conclus	ion
1.	Reference	ces



# **List of Figures**

Figure 1: RAWFIE iterative development process (2 <sup>nd</sup> cycle)	14
Figure 2: Gathering Information for Naval Search and Rescue (SAR) Operations scenario	20
Figure 3: UML Diagram for scenario 7	21
Figure 4: Visualization of resources waypoints	23
Figure 5:UML diagram for Scenario 8	24
Figure 6: Outlier detection for Scenario 8	25
Figure 7: RAWFIE Overall Component Architecture (see also [3], [2])	27



# List of Tables

Table 1: Abbreviations	13
Table 2: Exemplary Requirement Card used in this Deliverable	15
Table 3 List of Requirements Types	16
Table 4 List of subsystems and components	17
Table 5: Iteration 1 Requirements that remain valid and relate to Ethical issues	125
Table 6: Iteration 2 Requirements that relate to Ethical issues	125
Table 7: Overview of Iteration 2 defined requirements including traceability to	D3.1
Requirements	136
Table 8: Not mapped Requirements of iteration 1 and their status regarding RAWFIE system	m 137



## **Part III: Executive Summary**

The deliverable provides a deep look at the requirements and needs of the RAWFIE system. It attempts a more elaborated analysis and allocation of requirements to certain components based on experience gained and feedback provided during the 1<sup>st</sup> iteration cycle. RAWFIE deliverables D4.1 (High Level Design and Specification of RAWFIE Architecture) and D4.2 (Design & Specification of RAWFIE Components) were used as input as well as the experience obtained during the implementation and validation activities in the 1<sup>st</sup> year of the project. The use of the Slice Federated Architecture (**SFA**), considered mandatory for FIRE related projects was also taken into account during the requirement analysis.

The present document is the second in a series of three requirements analysis documents each one to be delivered in the beginning of each RAWFIE iteration cycle (see [1] 1.3.2. WT2 list of deliverables, page 93).



Part IV: Main Section



## **1** Introduction

## **1.1 Scope of Deliverable**

The purpose of this document, "D3.2 **Specification & Analysis of RAWFIE Components Requirements**", is to decompose the higher level requirements identified in D3.1 and assigning them to lower level functions (requirements allocation) as well as to identify new requirements that can be assigned to the various components of the RAWFIE architecture initially identified during the Platform Design activities of the first iteration cycle. The present document is the second deliverable in a series of three that will all focus on incrementally identifying requirements for the various RAWFIE components.

This document structure has as follows:

- Chapter 2 briefly restates the methodology adopted, the general formalizations followed and the templates used for recording requirements.
- Chapter 3 presents any updates and modifications that apply to the initial use cases (defined in the previous deliverable) while it defines some additional ones.
- Chapter 4 presents the result of requirement analysis performed in the second iteration. It records down the RAWFIE detail level requirements both functional and non-functional, following an appropriate categorization based on the defined components.
- Chapter 5 provides a traceability matrix between the initial user and system requirements (as defined in the 1<sup>st</sup> iteration) and the requirements defined in the present deliverable (2<sup>nd</sup> iteration)
- Chapter 6 provides a summary of the work performed in the present deliverable and sets the target for the next iterations

Abbreviation	Meaning
AHRS	Attitude and Heading Reference System
AGL	Above Ground Level
AP	Access Point
AT	Aerial Testbed
AUV	Autonomous Underwater Vehicle
<b>B-VLOS</b>	Beyond Visual Line Of Sight
CAA	Civil Aviation Authority
CAO	Cognitive-based Adaptive Optimization
CBNR	Chemical Biological Nuclear Radiological
СЕР	Circular Error Probability
CPU	Central Processing Unit
DETEC	Department of the Environment, Transport, Energy and Communication

## **1.2** Abbreviations

DGCA	Directorate General of Civil Aviation
DoA	Description of Activities
DoW	Description of Work (synonym to DoA)
EASA	European Aviation Safety Agency
ECC	Error Correction Code
EDL	Experiment Description Language
EU	European Union
E-VLOS	Extended Visual Line Of Sight
FIRE	Future Internet Research & Experimentation
FOCA	Federal Office of Civil Aviation
FPS	Frames Per Second
FPV	First Person View
GAA	German Aviation Act
GIS	Geographical Information System
GNSS	Global Navigation Satellite System
GPIO	General Purpose Input/Output
GPS	Global Positioning System
HD	High Definition
HW	Hardware
IAA	Irish Aviation Authority
IaaS	Infrastructure as a Service
IFR	Instrument Flight Rules
IDE	Integrated Development Environment
IP	Internet Protocol
ISO	International Standards Organization
JSON	JavaScript Object Notation
KPI	Key Performance Indicators
LBL	Long Baseline
MEMS	MicroElectroMechanical System
MM	Monitoring Manager
MSO	Multi Swarm Optimization
MT	Maritime Testbed
NF	Non Functional
OEDL	OMF EDL
OMF	Control and Management Framework
OS	Operating System
OTA	Over The Air
P2P	Point to Point
PSO	Particle Swarm Optimization
PTZ	Pan Tilt Zoom
RC	Radio Controller
RE	Requirement Engineering
RIA	Research and Innovation Action
ROS	Robot Operating System
ROV	Remotely Operated Vehicle
RPA	Remotely Piloted Aircraft
RPAS	Remotely Piloted Aircraft System
RPS	Remotely Piloted Station

SaaS	Software as a Service
SQL	Simple Query Language
TM	Testbed Manager
TMS	Testbed Manager Suite
TP	Testbed Proxy
UAV	Unmanned Aerial Vehicle
UGV	Unmanned Ground Vehicle
UI	User Interface
USB	Universal Serial Bus
USV	Unmanned Surface Vehicle
UxV	Unmanned System (of any type)
VFR	Visual Flight Rules
VLL	Very Low Level flight, below 150m above ground level
VLOS	Visual Line of Sight
VT	Vehicular Testbed
XML	Extensible Markup Language

Table 1: Abbreviations



## 2 Methodology

## 2.1 General

The methodology adopted has been described in the first version of the deliverable, thus it will not be analysed in details again. In brief we restate here that the overall requirements analysis activities are performed in the context of the RAWFIE iterative development process. Therefore, requirements in RAWFIE are defined incrementally since having a complete requirement specification from the very beginning is pretty difficult due to the inherent system complexity and the fact that certain constraints or issues are not evident until development activities start or even a first version of the system is put in operation.

The present work forms the basis of the second iteration cycle (see Figure 1). Although, this is not clearly depicted in the figure below, the second iteration partially overlaps with the first one. Feedback from design and development activities is used to modify, enhance and further refine the previous requirement specification.



Figure 1: RAWFIE iterative development process (2<sup>nd</sup> cycle)

The requirements process is comprised of four (4) main activities: requirements discovery, classification, prioritization and negotiation. During classification of requirements, coherency among requirements is achieved by organizing them according to the identified classification categories. Subsequently, prioritization and negotiation of requirements assists in identifying and resolving requirements conflicts.

Finally, we remind the use of the VOLERE methodology through the use of an appropriate template card for presenting requirements. Given the fact that in this second version of the



deliverable we focus more in system and component level requirements based on the 1<sup>st</sup> version of identified RAWFIE architecture we introduce a modification of the "requirement card" used to include also the name of the component or system where the requirement should be allocated. The {classId} value should adhere to the name of the component or system (a complete list of the available components is given in Table 4).

The type of requirement is based on the classification performed in the previous version of the deliverable (see Table 3)

Id:	{ClassId}-{XXX}	Туре:	follow categorizati on proposed in Volere template (see Table 3)	Importance (priority):	LOW, MEDIUM, HIGH	Source:	Requirement origin e.g.: Consortium Know-how, members, law regulation, standards, Deliverable. Iteration 1 Exp, etc.	Ver:	2
Title:         Requirement title/name (1 short sentence)									
Desc	<b>Description:</b> More detailed description of particular requirement (textual form only). If Requirement title is sufficient enough to understand the requirement, the can remain empty.				this fi	eld			
Addi (com	itional Info ments):	Any ac possible	lditional info e).	o to better cla	arify or illu	strate cor	acepts (pictures	s may	be
Com Subs	ponent or ystem	The co inferre	omponent or d also by Re	r subsystem equirement II	the requirer	nent is a	assigned to (s	hould	be
Refi	nes/Replaces	Should versior	Should be completed for requirements that modify, replace, or refine version 1 requirements						

 Table 2: Exemplary Requirement Card used in this Deliverable

Functional	Functional	FUNC		
	Data	DATA		
Non-functional:	Look and Feel Requirements	L&F		

Usability Requirements	USE
Performance Requirements	PERF
Operational - Environmental Requirements	ENV
Maintainability and Support Requirements	SUP
Security & safety Requirements	SEC
Other	OTH

Table 3 List of Requirements Types

Subsystem	ClassId	Component	Component ClassId	
General	GEN			
		General	PT-GEN-R	
		Web Portal	PT-WEB-P	
		Booking Tool	РТ-ВОК-Т	
		System Monitoring Tool	PT-SYM-T	
		Resource Explorer Tool	PT-REE-T	
		Experiment Authoring Tool	PT-EXA-T	
Platform	РТ	Experiment Monitoring Tool	PT-EXM-T	
		UxV Navigation Tool	PT-NAV-T	
		Visualisation Tool	PT-VIS-T	
		Data Analysis Tool	PT-DAA-T	
		Testbeds Directory Service	PT-DIR-S	
		EDL Compiler and Validator	PT-EDL-S	
		Experiment Validation Service	PT-EXV-S	
		Users & Rights Service	PT-USR-S	

		Booking Service	PT-BOK-S
		Launching Service	PT-LAU-S
		Visualisation Engine	PT-VIS-S
		Experiment Controller	PT-EXP-C
		Data Analysis Engine	PT-DAA-S
		System Monitoring Service	PT-SYM-S
		Accounting Service	PT-ACC-S
		General	TB-GEN-R
Testbed		Monitoring Monogon	
	ТВ	Monitoring Manager	TB-MOM
		Network Controller	TB-NEC
		Resource Controller	TB-REC
		Testbed Proxy	TB-PRO
		Testbed Manager	TB-MAN
		General	UXV-GEN
Testbed		UxV Node	UXV-NOD
		UxV Network and Communication	UXV-NET
UxV	UXV	UxV Sensor and Localisation	UXV-SEN
		UxV On-board storage	UXV-STO
		UxV On-board processing	UXV-PRC
		UxV Management	UXV-MGT

Table 4 List of subsystems and components



## 2.2 Definitions

To enable better formalization of requirements throughout this document, the following wording is encouraged to be used during definition of requirements:

**"Shall"** statements are binding requirements. They describe something that is mandatory. If a requirement uses "shall", then that requirement must be satisfied without fail. Non-compliance is not allowed. Failure to comply with one single 'shall' is sufficient reason to reject the entire product

"Should" is weaker. It can be regarded as a non-mandatory provision. It describes something that might not be satisfied in the final product, but that is desirable enough that any non-compliance shall be *explicitly* justified. Any use of "should" should be examined carefully, as it probably means that something is not stated clearly. If a "should" can be replaced by a "shall" or can be discarded entirely, so much the better.

**"May"** statements are also non-mandatory provisions. It grants permission to do something, and makes only a weak statement. It does not mean that it is possible to do it, only that you have permission to do it. In a user requirements document it shall only appear rarely, if ever. It is more appropriate to the detailed design where it could be used to define the behaviour of the product.

"Will" statements are non-mandatory, either they imply intent on design constraints or future tense.



## **3** User Scenarios

In the previous version of the deliverable six main scenarios were defined and used as a starting point to identify the user level and overall system level requirements. These scenarios included:

- Scenario 1 Environmental Monitoring of Water Canal
- Scenario 2 Border Surveillance or Perimeter protection of large area
- Scenario 3 On demand deployable Internet facilities
- Scenario 4 Exploration & Assessment of Network Technologies Robustness
- Scenario 5 Efficient Coordination for phenomena or mission coverage
- Scenario 6 Over the Air (OTA) UxV Re-programming

All these scenarios are still valid for iteration 2 while two extra scenarios have been identified and included in the list. Details on these scenarios are presented below.

## 3.1 Scenario 7 – Gathering Information for Naval Search and Rescue (SAR) Operations (ops).

## Overview/Rationale

In this use case RAWFIE platform will be used to mobilize resources that can collaborate for the purpose of gathering information for naval search and rescue (SAR) operations (ops). The potential environments of this scenario are wide sea area or sea area between islands, with intense coastline variations.

Potential end users for this scenario are:

- Governmental Organizations responsible for SAR operations.
- Non Governmental Organizations aiding SAR operations
- UAV, USV providers.

Picture



Figure 2: Gathering Information for Naval Search and Rescue (SAR) Operations scenario







#### Analytic Description

According to the scenario (Figure 2) the operational center, responsible for the SAR Ops of the BLUE sea area, receives information for a sinking vessel in the sea between the green Left and Right island areas. Immediately the central ops center gives order to the Regional Ops Centre (ROC) for spotting the vessel in danger. The RAWFIE platform, which controls a swarm of UAVs and USVs, loaded on a patrolling boat, mobilize the appropriate UxVs in accordance with the searching area. If the signal came from wide sea area, a swarm of UAVs is launched. If the area is near the coastline of the islands, USVs are launched. Finally, if the signal information is not accurate, a combination of surface and areal UxVs is used.



In any of the above cases, one UxV from the swarm is playing the role of the "mother" vehicle and is responsible:

- To guide the swarm effectively in order to scan the area of interest fast and thoroughly by utilizing efficiently the consumption of the limited resources.
- $\circ$  To gather the information from the swarm's sensors and transmit them back to the ROC.

When the vessel in danger is spotted, the ROC initiates the SAR operation and the UxVs return back to the patrolling boat.

In the context of the above described use case, RAWFIE platform can be utilized to execute a series of experiments in order to assess and identify the optimum way to utilize available UxV's resources in order to perform the requested task of scanning a specific sea area and spot an "object" in danger. Indicatively, the following experiments can be performed:

- Optimization of the used UxVs in relation to the particularities of the scanned sea area.
- Optimization of the UxVs search pattern.
- Optimization of the UxV's sensors used in relation to the requested information.

Type of sensors on the UxVs could be:

- o Day/night thermal cameras
- Radars
- o Sonars
- Acoustic sensors

# **3.2** Scenario 8 – Mobilize resources and gather sensor data (1st year review scenario)

#### Overview/Rationale

This section describes the scenario that was demonstrated during the 1<sup>st</sup> year project review. The main purpose of this scenario is to show the functionalities provided by the RAWFIE platform as a result of the 1<sup>st</sup> implementation cycle.

**Location:** UPTEC – Polo do MAR, Matosinhos

Date: 29 February 2016



#### UxV resources: USV and UGV

The experimenter uses the RAWFIE platform to define and run a simple experimentation scenario. In this use case the RAWFIE tools are utilized to mobilize the resources and retrieve measurements from the sensors. The sensor data that could be gathered during the experiment execution are: angular velocity, CPU usage, fuel usage, linear velocity, resources location, storage usage, system information, and voltage.

#### Picture



Figure 4: Visualization of resources waypoints







## Analytic Description

The experimenter through the RAWFIE platform is able to move the resources and gather a variety of sensor data. The steps of this scenario are the following (see also Figure 5 UML diagram):

- Login through the RAWFIE web portal
- The experimenter can check the available testbeds and resources
- In the next step the experimenter can define and validate an experimentation scenario
  - a. Authors an EDL script
  - b. Validate the experiment
  - c. Store the experiment for future launching
- Experiment launching
  - a. The experimenter can launch the experiment right after the definition
  - b. The experimenter can launch a stored experiment through the database
- During the experiment execution the experimenter is able to:
  - a. Visualize the resource waypoints (Figure 4)



- b. Gather sensor measurement
- c. Perform outlier detection through the data analytics tools (Figure 6)



Figure 6: Outlier detection for Scenario 8



## 4 System & Component Requirements

On the grounds of the first version of the requirements deliverable (D3.1) the pre-requirements analysis activities involved the following:

- Study Architecture Definition and defined components (mainly from D4.1 & D4.2)
- State of the art survey related to certain key elements of the RAWFIE system (EDL, UxVs)
- A more mature look to related FIRE projects
- A more detail look at testbed related to:
  - Deployment requirements and actions
  - o Specific HW & SW needed

The classification of requirements is based on information regarding the subsystems and components defined during the first iteration of architecture design. This information is presented in Table 4 and is also used within this section to provide appropriate sub-sectioning. Figure 7 provides an overview of the envisaged architecture components as defined in D4.1.



Figure 7: RAWFIE Overall Component Architecture (see also [4], [2])



## 4.1 Platform Requirements

The term Platform refers to the middleware solution responsible for managing and monitoring the lifecycle of an experiment in the context of the RAWFIE system. In the 1<sup>st</sup> iteration requirements were defined according to the experiment's lifecycle phases that included: authoring, booking, launching and evaluation of an experiment. Based on them a number of conceptual components were defined by WP4 (Design Phase).

#### 4.1.1 General

Id:	PT-GEN-R-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title	:	RAWFIE Platform should adopt Sliced Federated Architecture (SFA)							
<b>Description:</b> The RAWFIE Platform should be compatiliadopted in other FIRE projects. SFA prescrifederation of testbeds with different technologitadiministrators.						le with the bes a mini- ogies and	e overall SFA imal interface belonging to	to ena differ	ept ble ent
Addi (com	itional Info ments):								
Com	ponent or								
5403									
Refi	nes/Replaces	PT-P-0	01, PT-NF-0	08					

Id:	PT-GEN-R-002	Туре:	FUNC	Importance (priority):	HIGH	Source:	DoW	Ver:	2
Title	:	RAWFIE platform shall support various roles with different privileges at evel of access.							ery
Desc	cription:	The pla Every p exist: • Each of Definiti	atform shall blatform user Experiment Admin Testbed Op them provid	provide a set should be ass er erator ling different a pnal roles shou	of different ign to a role. ccess rights t ld be possible	roles wit At least to the vari e.	h predefined p the following r ous platform so	orivileg oles sh	;es. 1all

Additional Info	
(comments):	
Component or Subsystem	Web Portal, SFA interface
Refines/Replaces	PT-GEN-002

Id:	PT-GEN-R-003	Туре:	DATA	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2	
Title	:	The RA exchange	The RAWFIE Data model should include all basic entities that are used or/and exchanged by the various components of the RAWFIE Platform							
Description:       Such entities are:         •       Users         •       Resources         •       Testbeds         •       Experiments         •       Sensors         An exhaustive list should be defined definition and implementation docume				ed in the	appropriate co	ompon	ent			
Addi (com	tional Info ments):									
Com Subs	ponent or ystem									
Refi	nes/Replaces	PT-P-0	05							

Id:	PT-GEN-R-004	Туре:	DATA	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title	::	RAWF needs to the vari	IE platform to be persiste ous tools and	shall provide d, or used afte l services).	appropriate er an experin	data stora nent comp	age for inform bletion (e.g. an	ation t alysed	hat by
Desc	cription:	<b>ption:</b> The platform shall provide database services in the form of relational object database), that can be used for persisting information used or ex among the various services and tools. The exact information will be ba defined data model and may include:							l/or ged n a

	<ul> <li>Data information</li> <li>Spatial information</li> <li>Configuration information</li> <li>Historical information</li> </ul>
Additional Info (comments):	The database structure should adhere to the defined RAWFIE data model
Component or	
Subsystem	
Refines/Replaces	PT-P-005

#### 4.1.2 Web Portal

Id:	PT-WEB-P-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	DoW	Ver:	2
Title:         A web portal interface shall be provided to the users almost all main functionalities.							f the platform	to acc	ess
Desc	cription:	The RA (GUI), services	WFIE web acting as a s used by the	portal shall pr central point experimenters	ovide a user of access to s.	-friendly ( all the 1	Graphical User necessary reso	Interfa urces a	ace
Addi (com	tional Info ments):								
Com Subs	ponent or ystem	Web Po	ortal						
Refi	nes/Replaces	PT-GE	N-001						

Id:	PT-WEB-P-002	Туре:	FUNC	Importance (priority):	HIGH	Source:	DoW	Ver:	2
Title	2:	Web portal usage shall be allowed only to authenticated users							
Desc	cription:	An exp portal ( after th Single s	erimenter sh initial sign u e informatio sign-in authe	ould firstly be up). Access to on is reviewed ntication (login	registered b the portal fu and approv n) process sh	y creating inctionalit red by a ould be pr	an account they shall be allo RAWFIE adm rovided.	rough wed o inistra	the nly tor.

Additional Info	
(comments):	
Component or Subsystem	Web Portal, User & Rights Service
Refines/Replaces	PT-GEN-003

Id:	PT-WEB-P-003	Туре:	FUNC	Importance (priority):	HIGH	Source:	DoW	Ver:	2
Title:         A tutorial or similar type of documentation shaplatform							vided to the us	ers of	the
<b>Description:</b> A self-contained didactic material shall be provided to the experimenters the experiment design,, the use and the variety of resources, the testbed fact etc. This can be in the form of a wiki. These functionalities shall be availa all possible future experimenters that may be interested in RAWFIE fede and want to explore its capabilities						ters ab faciliti vailable federat	out ies, e to ion		
Addi (com	tional Info ments):								
Com Subs	ponent or ystem	Web Po	ortal (Wiki pa	age)					
Refi	nes/Replaces	PT-P-0	02						

## 4.1.3 Booking Tool

Id:	РТ-ВОО-Т-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title	:	Booking Tool should allow booking of resources at the experimenter level specified period and for selected resources							or a
Desc	cription:	Throug Reserva level of is a pre experin	h the bookin ation that inc reservation requisite for hent.	ng tool a pote cludes UxV re is related to a p proceeding wi	ntial experin sources from particular use th subsequer	nenter sho n one or 1 er only (us nt associat	ould be able to nore testbeds. er level reservation of resource	creat This f ation) a es with	e a irst and an



Additional Info (comments):	Reservation of resources are expected to be performed at 2 different levels (1) experimenter level and (2) experiment level (see also section 4.1.15 on Booking Service)
Component or Subsystem	Booking Tool
Refines/Replaces	PT-B-001

Id:	РТ-ВОО-Т-002	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title:Booking Tool functionality shall be compatible with the S architecture and the notion of slices reservations							vith the SFA	mysl	ice
<b>Description:</b> SFA and myslice implementation in particular, pro- reserving underline resources by allocating them in experimenter. RAWFIE booking functionality shall the functionality from there can fit its business model.						ovide mechar slices create try to reuse	nisms d by whate	for the ver	
Addi (com	tional Info ments):								
Com Subs	ponent or ystem	Bookin	g Tool						
Refi	nes/Replaces	PT-B-0	01						

Id:	РТ-ВОО-Т-003	Туре:	FUNC	Importance (priority):	HIGH	Source:	Arcitecture Deliverables	Ver:	2
Title	::	Booking Tool should delegate all its actions related to Booking of a resource the Booking Service							
<b>Description:</b> Booking Tool provides the front end of the overall reservation function logic related to actual booking, validations and interactions with should go via the Booking Service						vation function tions with the	1ality. Datab	Allase	
Addi	itional Info								
(con	iments):								
Com	ponent or	Bookin	g Tool						



Subsystem	
Refines/Replaces	PT-B-001

Id:	РТ-ВОО-Т-004	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title	:	Booking Tool shall also interact with the Testbeds Directory Service retrieve information on unallocated testbed resources							r to
<b>Description:</b> In order to provide the user/experimenter with a list of available resour initial reservation, the Booking tool shall retrieve information from the T Directory Service							ources Testb	for eds	
Addi (com	itional Info ments):								
Com Subs	ponent or ystem	Bookin	g Tool						
Refi	nes/Replaces	PT-B-0	01						

Id:	РТ-ВОО-Т-005	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title	:	Booking Tool should communicate with the underline services using JSON formatted messages (through an RPC or REST API)							
Description:									
Addi (com	tional Info ments):	The JS0	ON formattee	d messages use	d should res	pect the A	vro protocol		
Com Subs	ponent or ystem	Bookin	g Tool						
Refi	nes/Replaces	PT-B-0	01						

\_

\_

Id:	РТ-ВОО-Т-006	Туре:	FUNC	Importance (priority):	HIGH	Source:	Arcitecture Deliverables	Ver:	2			
Title:		Bookin reservat	Booking Tool should provide appropriate functionality for viewing the reservations of a user/experimenter									
Description:		An app with the The pag running experin	An appropriate page should be provided that enables viewing of Reservations with the involved resources. The page may include information on which resources are already involved in running or future scheduled experiments and provide means to navigate to the experiment info page									
Addi (com	itional Info ments):	An exp An adm	An experimenter should be able to see a list of Reservations made by him. An administrator should be able to view all users' reservations									
Com Subs	ponent or ystem	Bookin	g Tool									
Refi	nes/Replaces	PT-B-0	PT-B-002									

Id:	РТ-ВОО-Т-007	Туре:	FUNC	Importance (priority):	HIGH	Source:	Arcitecture Deliverables	Ver:	2			
Title:		Booking Tool should allow editing of existing Reservations										
Description:		<ul> <li>A user should be able to edit/modify existing reservations this may include:</li> <li>Modification of time reservation period (adding or removing timeslots)</li> <li>Adding and deletion of resources associated with existing reservation (only for resources not involved in running experiments)</li> </ul>										
Addi (com	itional Info ments):											
Component or Subsystem Booking Tool												
Refi	nes/Replaces	PT-B-0	02									

Id:	РТ-ВОО-Т-008	Туре:	FUNC	Importance (priority):	HIGH	Source:	Arcitecture Deliverables	Ver:	2
-----	--------------	-------	------	---------------------------	------	---------	-----------------------------	------	---

Title:	Booking Tool should allow cancellation of existing Reservations							
	Existing Reservations may be cancelled based on user request.							
Description:	• If no experiments are running or are associated with the reservation, a direct cancellation is possible.							
	• If running or scheduled experiments are found for a given reservation then the running experiments should allow to complete but the reservation should be marked cancelled and future scheduled experiments should be deleted.(or not allowed to be launched)							
Additional Info	A user should be able to cancel reservations created by him							
(comments):	An administrator should be able to cancel any reservation							
Component or	Desking Test Desking Comise							
Subsystem	BOOKING 1001. BOOKING Service							
Refines/Replaces	РТ-В-002							

Id:	РТ-ВОО-Т-009	Туре:	FUNC	Importance (priority):	HIGH	Source:	Arcitecture Deliverables	Ver:	2		
Title:		Booking Tool should allow creation of bookings through an intuitive UI interface									
Description:		<ol> <li>Booking tool should provide a step wizard that will permit:         <ol> <li>Definition of booking date and time via selection of discrete timeslot(s) on a Calendar and timeline view</li> <li>Selection of resources that will be included in the Reservation (only available resources for the timeslots defined in step 1 should be available)</li> <li>Issue a request for reservation</li> </ol> </li> </ol>									
Addi (com	tional Info ments):	In order to achieve step 2 experimenter should be able to retrieve infabout the testbeds and their resources. Since resources are whole UxV systems there is a possibility that t reservation response is not directly available. In such a case the experimental be informed via a proper notification mechanism							ion tual		
Com Subs	ponent or ystem	Bookin	g Tool								
Refi	nes/Replaces	PT-B-0	Г-В-002								

Id:	РТ-ВОО-Т-010	Туре:	FUNC	Importance (priority):	HIGH	Source:	Arcitecture Deliverables	Ver:	2		
Title:		Appropriate notification mechanism should be provided to the user in case status of reservation request is not directly available.									
Desc	cription:	Since resources are whole UxV systems there is a possibility that the actual reservation response is not directly available. In such a case the experimenter should be informed via a proper notification mechanism									
Addi (com	itional Info iments):	This may include situations where a reservation is impossible to be satisfied the future.							l in		
Com Subs	ponent or ystem	Bookin	g Tool, Bool	king Service							
Refi	nes/Replaces	PT-B-0	02								

Id:	РТ-ВОО-Т-011	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Other	Ver:	2			
Title:		Booking Tool may provide assistance of feedback to the potential experimenter during the booking process										
Description:		In order to facilitate the experimenter during the initial reservation of resources, the booking tool may provide information to the user regarding the booked resources per timeslot or the available timeslots per testbed.										
Addi (com	itional Info ments):											
Com Subs	ponent or system	Bookin	g Tool									
Refi	nes/Replaces	PT-B-0	05									

Id:	РТ-ВОО-Т-012	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title:		Booking booking	g functional g as well as p	ity should protect for male	ovide means evolent action	s to ensu ns that a u	re fairness in ser may perfor	resou m.	rce
Description:	<ul> <li>The booking process should ensure that some checks/validations apply to ensure resource reservation fairness and avoid spurious actions that may lock out other users like:</li> <li>Reservation of enormous size of resources by a single user</li> <li>Reservation of resources for very long lasting periods</li> </ul>								
--------------------------------	---								
Additional Info (comments):	The process may use some configurable max limits for number of resource, number of consecutive timeslots, total number of reservations that should be validated upon issuing a Booking request.								
Component or Subsystem	Booking Tool, booking Service								
Refines/Replaces	РТ-В-005								

Id:	РТ-ВОО-Т-013	Туре:	FUNC	Importance (priority):	LOW	Source:	Consortium	Ver:	2			
Title	:	RAWF.	RAWFIE platform should allow virtualization of available UxVs resources during reservation process									
Desc	Description: Unless an experimenter explicitly requests reservation of testbeds/resources for an experiment, the RAWFIE platform should of experimenter the ability to reserve resources in a topology agnostic mar offering virtualization of available resources. Internally the service should attempt to reserve resources in the same testbed and if this is not possible then consider resources from multiple At the same time the service has to guarantee that the reserved resourceally be available for the experiment.						spec: offer to anner the e physic testbe urces v	ific an hus ical ical ical will				
Addi (com	tional Info ments):	Exact le next ite	Exact level of virtualization that will be available in RAWFIE will be defined in next iteration of the deliverable									
Com Subs	ponent or ystem	Bookin	g Tool									
Refi	nes/Replaces	PT-B-0	06									

## 4.1.4 System Monitoring Tool

Id:	PT-SYM-T-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title	2:	Listing	and/or visua	lisation of curr	ent system h	ealth statu	is shall be avai	lable.	
Description:       The users of the RAWFIE platform shall be informed about the system status. This includes:         •       Hardware servers are up and running         •       Services (application server, message bus, databases) up and runnin         •       Testbeds are connected and ready         •       UxVs are connected and ready						em hea	ılth		
Add (con	itional Info ments):								
Com Subs	ponent or system	System	Monitoring	Tool					
Refi	nes/Replaces	(PT-NF	5-007)						

Id:	PT-SYM-T-002	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2
Title	::	The cur	rent system	health status sh	nould be grou	iped them	atically.		I
Desc	cription:	For bet under the The gro	ter compreh- ne componer ouping of con servers of th Testbeds UxV	nensibility all nt. mponent may b he cloud infrast	services of the by tructure	a compor	nent should be	e grouj	ped
Addi (com	tional Info ments):								
Com Subs	ponent or ystem	System	Monitoring	Tool					
Refi	nes/Replaces								



Id:	PT-SYM-T-003	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2
Title	:	Filterin be poss	g of the acce ible.	essible compor	nent health st	tatuses by	user roles/righ	its sho	uld
Desc	eription:	Based of should	on the access be filtered ou	s rights of the ut.	user, the hea	alth status	es of special co	ompon	ent
Addi (con	tional Info ments):								
Com Subs	ponent or ystem	System	Monitoring	Tool					
Refi	nes/Replaces								

\_

Id:	PT-SYM-T-004	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2
Title	2:	The hea	The health statuses webpage should be updated automatically.						
Desc	cription:	The cu System update	rrent health Monitoring interval shou	statuses shou Service and th ild be configura	ld be reque e webpage sl able	sted at fi nould be u	xed intervals pdated accordi	from ingly. 7	the Гhe
Addi (com	itional Info ments):								
Com Subs	ponent or system	System	Monitoring	Tool					
Refi	nes/Replaces								

Id:	PT-SYM-T-005	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title	:	The heater textual	alth status in information	formation sho with additional	uld include a details.	a severity	indication and	l possi	bly
Desc	cription:	The inf or seve Extra in	ormation rec rity field winformation r	reived for a co of the possible vanay be received	mponents he alues (CRIT) ed with addit	alth status CAL, W.	s should includ ARNING, NO ails regarding	e a sta RMAI the hea	tus 2). 11th



	status.
Additional Info	
(comments):	
Component or	System Manitoring Tool
Subsystem	System Monitoring 1001
Refines/Replaces	

## 4.1.5 Resource Explorer Tool

Id:	PT-REE-T-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	DoW	Ver:	2	
Title	:	The UI interface shall illustrate testbed and UxV information of the RAWFIE federation that the experimenters should take advantage of								
Description:       Essential information provided shall include at least:         • Testbed facilities information         • UxVs information										
Addi (com	itional Info ments):									
Com Subs	ponent or ystem	Resource	ce Explorer 7	Гооl						
Refi	nes/Replaces	PT-P-0	01, (PT-P-00	)3)						

Id:	PT-REE-T-002	Туре:	FUNC	Importance (priority):	LOW	Source:	Iteration1 Exp	Ver:	2		
Title	:	Registra	gistration of testbeds and UxVs may be possible via the Web Portal								
<b>Description:</b> Editing of all relevant information about testbeds and UxVs may be possible							sible.				
Addi (com	tional Info ments):	This fu discove	This functionality is needed if the testbed does not support automatic resource discovery.								
Com Subs	ponent or ystem	Resource	ce Explorer T	Fool							



Refines/Replaces	PT-P-004

Id:	PT-REE-T-003	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	2		
Title	:	Resource	ce Explorer t	ool shall allow	for fine-gra	ined resou	irces' searches				
		Resourd experin needed	ce Explorer nenter identi for an experi	tool shall pro fying testbea ment.	ovide basic o d or/and Ux	query cap V resour	abilities to fac ce specific ca	ilitate apabilit	the ties		
Desc	ription:	An experimenter shall be able to fill in some specific technical details about the hardware he/she is looking for. It should be possible for the resource discovery tool to construct a suitable response based on the resource information provided for a testbed.									
		When the query in the resource discovery phase returns a certain list of resources, it should be possible for the experimenter to select the resources that would like to include in the experiment. This should be supported in relation to a specific resource ID.									
Addi	tional Info	Need to define what exactly these capabilities could be for the testbed node and its various resources (i.e. CPU, RAM, Op. system, battery state, communication interfaces, sensor types, capabilities regarding resource controller, etc.)									
(com	nments):	Need also to agree whether query capabilities would be available via an SQL query like language or via appropriate drop down menus or catalogues (the latter might be preferable for novice users but may limit the complexity of queries and consequently the granularity of searches).									
Com Subs	ponent or ystem	Resource Explorer Tool									
Refi	nes/Replaces	PT-A-0	16								

Id:	PT-REE-T-004	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	2			
Title:		Link to	Link to the Booking Tool should be provided									
Description:		It shou Bookin	ld be possil g Tool with t	ble to book the selected rest	he found rest sources shoul	sources. S ld be prov	So links that of ided.	opens	the			
Addi (com	tional Info ments):											



Component or Subsystem	Resource Explorer Tool
Refines/Replaces	PT-P-001, (PT-P-003)

### 4.1.6 Experiment Authoring Tool

Id:	PT-EXA-T-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2		
Title:		Experir definiti	nent Descrip on of experir	otion Language nent scenarios	e (EDL) sha	ll be used	d as a languag	ge for	the		
Desc	cription:	A Dom known handlin	A Domain Specific Language combining some common characteristics of well- known scripting languages shall be developed for the effective creation and handling of simple or complex experiment scenarios.								
Add (con	itional Info ments):	The ED •	L shall provi Syntax Colo Content Ass o com synd o loca o don o elen Validation a	ide: oring sist nmon constru- chronization bl ation/topology nain specific el ments for descr and Quick Fixe	acts like locks, task de specific elen ements speci ibing the Ux	loops, c efinitions o nents ific to eacl V behavio	conditional st etc. n UxV testbed or	atemer	nts,		
Com Subs	ponent or system	Experin	nent Authori	ng Tool							
Refi	nes/Replaces	PT-A-0	01								

Id:	PT-EXA-T-002	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2	
Title	:	The EI experin	The EDL shall allow the definition of all necessary requirements for an experiment							
Desc	ription:	The exp	perimenter sh The number	nall be able to c	lefine for the	e available e experim	booked resour	ces:		

	• The name of the testbed
	• The initial position of the UxVs
	• The time duration of the experiment
	• The maximum distance that the UxVs can cover
Additional Info (comments):	These are important features for the setup of the resources and their usage during the experiment. These features also help to the validation phase.
Component or Subsystem	Experiment Authoring Tool
Refines/Replaces	PT-A-002

Id:	PT-EXA-T-003	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	2	
Title	:	For each defined experiment specific metadata, i.e. name, version, date and description shall be defined.								
<b>Description:</b> In RAWFIE experimenters that create an experiment will need to provide high-level description of the experiment via metadata and its purpose allows infrastructure providers to keep track of the usage of the infrast and enables them to report about this to their funding sources.							de a sh oose. T astruct	iort This ture		
Addi (com	tional Info ments):	Additio providi experin	nally searcl ng a short nent.	hing and reu description,	sing experin possibly ev	ments is zen a vi	greatly simp deo demonstr	olified ating	by the	
Com Subs	ponent or ystem	Experir	Experiment Authoring Tool							
Refi	nes/Replaces	PT-A-0	02							

Id:	PT-EXA-T-004	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	2		
Title	:	An exp parame	An experimenter shall be able to provide initial conditions and/or configuration parameters for an experiment								
Description:		The EI configu exhaust	DL should s ration param ive list): initial positi	support experi neters for an ex	menter in d periment. Su ources	efining in ch conditi	nitial condition ons may include	ns and de (not	d/or t an		

	<ul> <li>specific communication interface to be used</li> <li>the enabled sensors</li> <li>etc</li> </ul>
Additional Info (comments):	It should also be possible to define what happens if the initial conditions are not met (abort the experiment, run it with additional sensors needed to gather the initial situation etc.).
Component or Subsystem	Experiment Authoring Tool
Refines/Replaces	PT-A-009

Id:	PT-EXA-T-005	Туре:	FUNC	Importance (priority):	HIGH	Source:	Scenario	Ver:	2
Title	:	An exp during	erimenter sh experiment a	all be able to uthoring	manage/gui	de the ava	ilable booked	resour	ces
Desc	cription:	The exp	Waypoints of Timeline of predefined in Data manag Communica the experim Sensor man when	erform for each management: d nanagement (se intervals) gement – which ation managem enters nagement- which	n node/ a gro lefine specifi equential or a sensor will hent- which n ch sensor w	up of node c waypoir parallel send data network p ill be acti	es its at the opera execution, exe in a time interv rotocol will b vated / deacti	ting tra ecution al e used vated a	at by
Addi (com	tional Info nments):	Trigger constrait triggere Type of support	Triggered based activation can be initiated based on the fulfillment of certain constraints (i.e. battery below a certain level). The constraints supported for triggered based activation/deactivation are still to be defined. Type of events may relate to a failure or malfunction (or other criteria). List o supported events is still to be defined.						
Com Subs	ponent or ystem	Experir	nent Authori	ng Tool					
Refi	nes/Replaces	PT-A-0	04						

Id:	PT-EXA-T-006	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2	
Title		An experimenter shall be able to define the type of information to be gathered and/or stored by UxV resource(s)								
Desc	cription:	During UxV re should gathere	During experiment authoring the experimenter should be able to prescribe for a UxV resource the type and characteristics of the (sensor's) information the should be gathered in a specific time interval The types of the informatio gathered will be proposed to the experimenter as an auto-complete function.							
Addi (com	itional Info iments):									
Com Subs	ponent or ystem	Experir	nent Authori	ng Tool						
Refi	nes/Replaces	PT-A-0	06							

Id:	PT-EXA-T-007	Туре:	FUNC	Importance (priority):	HIGH	Source:	Scenario	Ver:	2		
Title	:	An exp stored c	An experimenter shall be able to define the type of metrics to be gathered and/or stored during an experiment and/or per UxV resource								
During experiment authoring the experimenter shou metrics or performance indicators that need to be c analysis. These metrics may include:         • network related metrics (i.e. distributions of etc) (check scenario 4)         • energy/consumption related metrics (i.e. information quality metrics (i.e. information 5)						should b be collections of e s (i.e. nation free	e able to defin cted and stored rrors, SNR, th coverage vs eshness) (check	e speci l for la roughp ener c scena	ific ater out, rgy ario		
Addi (com	tional Info ments):	The EI addition	DL should shal ones iden	support some tified in the fur	basic type o ture.	of metric	s and be exte	endable	if		
Com Subs	ponent or ystem	Experir	nent Authori	ng Tool							
Refi	nes/Replaces	PT-A-0	07								

Id:	PT-EXA-T-008	Туре:	FUNC	Importance (priority):	HIGH	Source:	Scenario	Ver:	2		
Title	:	An exp during o	An experimenter shall be able to provide navigation or movement directives during experiment authoring								
<b>Description:</b> The EDL should provide the capability to define navigative directives. This can be done in the form of geo-reference predefined movement patterns when group of nodes is the case							avigation or 1 eferenced way e case.	novem <sup>a</sup> points	ent or		
Addi (com	tional Info ments):	Experin map the	nent authori e points of ea	ng tool will p ach node in ord	rovide a ma er to define a	p. Experin a specific	nenter can ma trace.	irk on	the		
Com Subs	ponent or ystem	Experin	nent Authori	ing Tool							
Refi	nes/Replaces	PT-A-0	08								

Id:	PT-EXA-T-009	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Scenario	Ver:	2
Title	:	An exp specific	n experimenter should be able to create groups of UxVs resources, fo becific directives will apply.						
Desc	cription:	directives that a group of UxV resources should follow during an experimexecution.							ion ent
Addi (com	itional Info ments):	Formati MSO) t	Formation info may be provided in the form of certain algorithms (i.e. PSMSO) that should be adopted by the UxVs for their optimal placement						
Com Subs	ponent or ystem	Experin	Experiment Authoring Tool						
Refi	nes/Replaces	PT-A-0	10						

Id:	PT-EXA-T-010	Туре:	FUNC	Importance (priority):	HIGH	Source:	DoW	Ver:	2		
Title:		A textu	A textual editor shall be provided for the authoring of RAWFIE experiments								

Description:	A textual editor tool providing access to all EDL elements and all the functionality needed to edit experiment scenarios shall be provided.
Additional Info (comments):	Ideally the editor will be an IDE with a code completion, syntax highlighting, syntax checking, debugging capabilities as well as other features making the authoring process easier and more productive.
Component or Subsystem	Experiment Authoring Tool
Refines/Replaces	PT-A-011

Id:	PT-EXA-T-011	Туре:	FUNC	Import (priorit	tance ty):	HIGH		Source:	DoW		Ver:	2
Title	:	A visu experin	visual/graphical editor shall be provided for the authoring of R aperiments							RAWI	FIE	
Desc	cription:	The visual editor tool shall provide a graphical interface for ha experiments.							handl	ing		
Addi (com	itional Info nments):	-										
Com Subs	ponent or system	Experir	nent Authori	ing Tool								
Refi	nes/Replaces	PT-A-0	12									

Id:	PT-EXA-T-012	Туре:	FUNC	Importance (priority):	HIGH	Source:	Other	Ver:	2	
Title	:	Platforr EDL	Platform shall allow saving, editing and/or deletion of an experiment defined EDL							
<b>Description:</b> The experimenters shall have the option to save an experiment and re later on demand. They shall also be allowed to delete or modify scenarios owned by them. Every version of the scenario will be saved and retrieved later on, i.e. an integrated version control system will be available is done by assigning a unique ID to every saved scenario version and s database						retrieve v exist ind can able. T stored	e it ing be his l in			



Additional Info	-
(comments):	
Component or Subsystem	Experiment Authoring Tool
Refines/Replaces	PT-A-015

Id:	PT-EXA-T-013	Туре:	FUNC	Importance (priority):	HIGH	Source:	Other	Ver:	2
Title	:	The vi waypoi	sual editor nts from a m	should allow ap	the definit	tion of n	novement and	l locat	ion
Desc	cription:								
Addi (com	itional Info nments):								
Com Subs	ponent or system	Experir	nent Authori	ing Tool					
Refi	nes/Replaces	PT-A-0	12						

Id:	PT-EXA-T-014	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2	
Title	:	During to the o	During authoring of an experiment selection of resources should be limited or to the ones previously reserved from the user at the foreseen time of experiment							
Desc	cription:	The selection of resources to be included in an experiment should be based of previous reservation performed by a user/experimenter. Only reserved resourt for the expected time of experiment should be available for inclusion in the E script.						n a ces DL		
Addi (com	itional Info ments):	After the somehous exclude	After the inclusion of a resource to an experiment script, the resource shoul somehow be flagged as reserved for the experiment timeslots in order to b excluded from future experiment definitions.							
Com Subs	ponent or system	Experin	xperiment Authoring Tool							



Refines/Replaces	

Id:	PT-EXA-T-015	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2	
Title	:	Validation of EDL script should be possible prior to or during saving								
Desc	ription:	on: Additional contextual validations may apply during the saving process (pos by contacting the Validation Service)						on dur (possi	ing bly	
Addi (com	tional Info ments):	Each ex	periment sho	ould be valid in	n syntax, sem	nantics and	1 securiry cons	traints.		
Com Subs	ponent or ystem	Experir	Experiment Authoring Tool							
Refi	nes/Replaces	PT-L-0	02							

Id:	PT-EXA-T-016	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Scenario	Ver:	2	
Title	itle: An experimenter shall have the means to define actions on a periodic or ad hoc basis during execution of an experi							hould	run	
Desc	cription:	<ul> <li>The EDL should support the definition of actions or sequence of actions (tash that may run periodically or triggered based on predefined criteria or even Such actions may related to: <ul> <li>enablement/disablement of certain functionality (or modules)</li> <li>data storage (or caching)</li> <li>data transmission</li> <li>error reporting</li> </ul> </li> <li>Additional type of actions may exist based on scenario specific needs</li> </ul>							ks) nts.	
Addi (con	tional Info ments):									
Com Subs	ponent or ystem	Experir	Experiment Authoring Tool							



Refines/Replaces	PT-L-010

## 4.1.7 Experiment Monitoring Tool

Id:	PT-EXM-T-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	DoW	Ver:	2
Title:         A RAWFIE user should be able to view an overview of his/he						is/her experim	ents		
<b>Description:</b> A user will be provided with a page showing his/her experiments (frunning or scheduled ones). Also the monitoring tool shall mana presentation of the information needed for monitoring the status of the during the experiments.						(finish inage the noo	ed, the des		
Addi (com	tional Info ments):								
Com Subs	ponent or ystem	Experin	nent Monitor	ring Tool					
Refi	nes/Replaces	PT-L-0	04						

Id:	PT-EXM-T-002	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2
Title	:	Experin	Experiment Monitoring and Visualisation should be integrated						
Desc	cription:	tion: The values of the sensing modules and the status of the different network modules are some essential elements of the monitoring process.							ing
Addi (com	itional Info ments):	The vis tools sh	ualisation of ould work to	f collected data	a is done via	a the Visu	alisation Tool.	. The t	wo
Com Subs	ponent or ystem	Experin	Experiment Monitoring Tool, Visualisation Tool						
Refi	nes/Replaces								



Id:	PT-EXM-T-003	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2	
Title	:	Cancellation of running experiments should be possible via Web Portal								
Desc	cription:	A runni serious	ng experime problems.	ent should be a	ble to be car	ncelled if t	he experiment	er noti	ces	
Add	tional Info									
(con	nments):									
Com	ponent or	<b>.</b> .		·	1					
Subs	ystem	Experin	nent Monitor	ring 1001, Boo	king Tool					
Refi	nes/Replaces									

### 4.1.8 UxV Navigation Tool

Id:	PT-NAV-T-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	DoW	Ver:	2
Title:This component will provide to the user the ab of UxVs through a user friendly interface.						oility to re	motely navigat	te a squ	uad
Through a user friendly interface, the experimenter details of the experiment, providing information reg vehicles, the number of the units etc.						menter w ion regar	vill specify the ding the numb	e requi per of	red the
Desc	cription:	Navigat extensit the abil to perf constru	ting an UxV ve training to ity to non-ex form basic ction, area su	' is not an eas become prof pert users to re navigation m urveillance and	y task and re icient. The U emotely guid issions such path plannir	equires in JxV Navi e a squad 1 as way 1g.	itial instruction gation Tool wi of robotic vehi point navigat	ns and ll prov icles so ion, n	an ride ) as nap
Addi (com	itional Info iments):	The vir turn bas Throug for eacl Control	The virtual controller will allow the experimenter to guide the vehicles using a turn based navigation mechanism and to collect data from their equipped sensors. Through the provided interfaces Rawfie users specify the next desired location for each unit. In the sequel, these instructions are transmitted to the "Experiment Controller" and sequentially are translated, evaluated and delivered to the robots.						
Com Subs	ponent or system	UxV N	avigation To	ol					
Refi	nes/Replaces	PT-L-0	08						



Id:	PT-NAV-T-002	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2	
Title		The tool should provide some validation of user's instructions								
Desc	cription:	The Uz validati	xV Navigati on mainly re	ion Tool comp elated to legalit	oonent shou y of provided	ld provide d instructio	e some basic ons.	real ti	me	
Additional Info (comments):Each experiment should compatible with the resource controller.						ntroller.				
Com Subs	ponent or system	UxV Na	avigation To	ool						
Refi	nes/Replaces									

Id:	PT-NAV-T-003	Туре:	FUNC	Importance (priority):	HIGH	Source:	DoW	Ver:	2	
Title	:	UxV N resourc	lavigation T es	ool should be	e available f	for the na	vigation of a	ll mov	ing	
Desc	cription:	Real time navigation may be restricted by the communication technology of the UxV data transmission.								
Addi (com	itional Info nments):									
Com Subs	ponent or system	UxV N	avigation To	ool						
Refi	nes/Replaces	PT-L-0	08							

Id:	PT-NAV-T-004	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2	
Title:		UxV Navigation Tool should be available to read from the database a detailed version of the map of the available areas								
Desc	ription:	A map of the area will illustrate the current position of each robot. Simply, by clicking on the map, the users define the next desired location								



Additional Info	
(comments):	
Component or Subsystem	UxV Navigation Tool
Refines/Replaces	

#### 4.1.9 Visualisation Tool

Id:	PT-VIS-T-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2	
Title	:	The Visualisation Tool shall allow the visualisation of information about the running experiments, in tabular/graphical form								
Desc	cription:	From the time" v the sum	the Visualisation isualisation of unary of the current loca values of al for the expe the value of experiment	ion Tool GUI, of the informat same informat tion (e.g. lat ar l measurement of any other l purposes	it should be ion coming t ion after the ad lon values ts coming fro kind of para	possible t from the e experimen ) of each n om the di- nmeters re	to access to "ne experiment, as nt stops. This in resource fferent sensors elevant for the	ear to r well as ncludes availa e spec	real s to s: ible ific	
Addi (com	tional Info ments):									
Com Subs	ponent or ystem	Visuali	sation Tool							
Refi	nes/Replaces									

Id:	PT-VIS-T-002	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	DoW	Ver:	2	
Title:		A 3D visualization should be available for the tracking of all moving resources								
Description:		Togethe executio availabl Visualis of 3D C	er with the tr on of the e le. The pos sation Tool SIS or image	racking of UxV experimentation ssibility that and the VIsual data sources, o	7 resources of n scenario, 3D visualization Engi of the experim	on a traditi 3D visua ation will ne will de ment and t	ional 2D map of lisation should be supporte epend on the a he experiment	luring l be a d by vailabi ation a	the Iso the lity rea	



Additional Info (comments):	Real time tracking may be restricted by the communication technology of the UxV data transmission. 3D visualization is possible only if suitable 3D maps of the area of interest will be available for free.
Component or Subsystem	Visualisation Tool
Refines/Replaces	PT-L-006

Id:	PT-VIS-T-003	Туре:	FUNC	Importance (priority):	LOW	Source:	Architecture Deliverables	Ver:	2		
Title	:	The Vi the exp	sualisation T eriment, and	disation Tool may allow visualisation of video streams coming from ment, and experiment's camera control							
Desc	cription:	From the Visualisation Tool GUI, it may be possible to get and visualize video streams coming from cameras on board of the devices or placed in the experiment's area. In such cases, a functionality could be provided so that the experimenters can control the position of the cameras directly from the web browser, by sending specific commands to the cameras/devices							deo the the veb		
Add (con	itional Info ments):	The fur experin	actionality w nent's locatio	ill be available on, or on board	provided th of the UxVs	at camera	s will be availa	able at	the		
Com Subs	ponent or system	Visuali	Visualisation Tool								
Refi	nes/Replaces										

Id:	PT-VIS-T-004	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2
Title	:	The Visualisation Tool shall provide access to information / features associate to each UxV device on the geographic map							
Desc	cription:	From tl associa Availat	ne Visualisat ted to each U ole information current loca list of on-bo current valu basic inform	tion Tool GUI JxVs, after e.g. on may include tion (e.g. lat ar oard sensors es of all measu nation about the	, it shall be clicking on e: nd lon values nrements con e status of th	possible t the specif ) ning from e device	to access to the constant of the different set	e featu 1 the m ensors	ap.



Additional Info	
(comments):	
Component or Subsystem	Visualisation Tool
Refines/Replaces	

Id:	PT-VIS-T-005	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2	
Title:		The Visualisation Tool shall allow organization and manipulation of multiple geographic layers								
Description:       The Visualisation Tool G geographic elements as oviinclude:         • UxVs themselves, a         • Specific, detailed to building (indoor)         • Other geo-referenc layers         It will be possible to show base map (e.g. Google Map.				Tool GUI sh s as overlays selves, and asso etailed maps of door) referenced inf to show / hide gle Map, Open	all allow to on the map ociated sense of the exper formation su the differen Street Map)	o add ar . Such ge ors timent are thas ro t layers, a	nd manipulate eographic elem ea (outdoor, in ads, obstacles as well as to c	multi ients n ndoor) , therr hoose	ple nay or nal the	
Addi (com	tional Info ments):									
Com Subs	ponent or ystem	Visuali	sation Tool							
Refi	nes/Replaces									

Id:	PT-VIS-T-006	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Architecture Deliverables	Ver:	2		
Title	:	Possibil	ity of Addin	g/Removing/U	pdating grap	hical wid	gets should be	provid	ed		
Description:		The experimenter can directly edits the widgets in the browser window. The new widgets are plotted on the screen. The user can adjust the information on the screen based, on the requirements and the current scenario.									



Additional Info	
(comments):	
Component or	
Subsystem	Visualisation Tool
Refines/Replaces	

Id:	PT-VIS-T-007	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2		
Title	:	Possibility to display both actual and expected UxVs' route and position should be provided									
Desc	cription:	As the path be and in g possibil expecte	position and tween waypo general for se lity to visuali d	the route of the pints, for the pr ecurity and safe ise the actual p	e UxVs may esence of ob ety reasons, t osition and r	change du stacles, ot he tool sh oute, as w	te to recalculat her UxVs in th ould provide th ell as the ones	ion of t le path, ne origina	the , ally		
Addi (com	itional Info ments):										
Com Subs	ponent or system	Visuali	sation Tool								
Refi	nes/Replaces										

#### 4.1.10 Data Analysis Tool

The Data Analysis Tool is the main UI interface that relays information to the Data Analysis Engine. It implements the standard UI decoupling interface. The Analysis Tool has three components: the data selection section, the result visualization (via graphite) and the job manager which is provided via the Spark jobserver.

Id:	PT-DAA-T-001	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2		
Title:		Analysis tool will provide interface to data engine.									

Description:	The Data Analysis Tool provides a user interface with which the consumer can select data metric(s) and a data analytics procedure, coupled with source and destination points. This information is relayed to the analytics engine which builds the required spark job.
Additional Info (comments):	<ul> <li>The metrics that will be support are currently restricted to integer/floating point values, however a user may decide to write a custom job that utilizes character values for say NLP.</li> <li>A spark job is basically a model coupled with the parameters for it. For clarity we will refer to the model as the spark 'jar' and the parameters + model as the spark 'job'</li> </ul>
Component or Subsystem	Data Analysis
Refines/Replaces	PT-E-003, PT-E-002

Id:	PT-DAA-T-002	Туре:	FUNC	Importance (priority):	LOW	Source:	Iteration1 Exp	Ver:	2
Title	Title:         Analysis tool will provide access to past experiments								
Desc	ription:	Access databas	will be prov e that holds <sub>J</sub>	ided to referen previous result	ce past expe s	riments &	results via a ti	ime ser	ries
Addi (com	tional Info ments):	Every e	xperiment sł	hould be uniqu	ely identified	l within th	e RAWFIE pla	atform	
Com Subs	ponent or ystem	Data A	nalysis						
Refi	nes/Replaces	PT-E-0	03, PT-E-00	1					

Id:	PT-DAA-T-003	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2	
Title:		Analysis tool will provide ability to query message bus streams								
Description:		Using t availab	he Jobserver le streams an	r UI interface ad metrics	the analysis	tool show	uld be able to	query	all	

Additional Info (comments):	<ul> <li>Our use case for our provided jobs currently restrict jobs to one job per metric. The metric restriction definition is provided in PT-DAA-T-001</li> <li>Messages between the analysis tool and engine will take place via a simple message exchange on the message bus.</li> <li>The definition of this schema is provided in WP5         <ul> <li>It encompasses a model structure as well as source and destination location</li> </ul> </li> </ul>
Component or Subsystem	Data Analysis
Refines/Replaces	PT-E-004

Id:	PT-DAA-T-004	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2	
Title	:	Analysis tool will provide interface to end running jobs								
Desc	cription:	If a job is stuck or the user wants to abort, there should be an interface to be able to do this.								
Addi (com	tional Info ments):	This function the Ana well as	nctionality is Ilysis Tool. links to all t	s provided by 1 The job server he logs on the	inking native shows all th workers.	ely to the S ne running	Spark Jobserve g jobs, the faile	er UI fr ed jobs	om ; as	
Com Subs	ponent or ystem	Data Ai	nalysis							
Refi	nes/Replaces	PT-E-0	04, PT-E-00	3						

Id:	PT-DAA-T-005	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2	
Title:		Analysis tool will provide a simple metric selection interface, a view of the result stream & the job status tab								
Description:		The Da desired Finally, PT-DA	ta Analysis ' metric. It v it will also A-T-004	Tool will prov will also have have a tab to t	ide a simple a tab to gr he jobserver	list based aphite to UI which	l system of sel- view the resu is talked abou	ecting lt strea it more	the am. e in	

Additional Info (comments):	
Component or Subsystem	Data Analysis
Refines/Replaces	PT-E-003, PT-E-002

## 4.1.11 Testbeds Directory Service

Id:	PT-DIR-S-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2		
Title	:	The Testbed Directory Service shall provide access to information on all Testbeds registered in RAWFIE									
Desc	cription:	The Testbed Directory Service shall provide the Web Service interface for other RAWFIE components to be able to access information on the testbeds' registered in the RAWFIE database.									
Addi (com	itional Info ments):	Provide nan geo sho usa type tota list con cap	d testbeds' in graphic loca rt descriptio ge) e of resource l number of of resources nectivity / he abilities in te	nformation inc tion on (possibly m es supported/av resources avail with an indica ealth status erms of availab	ludes: entioning gu ailable lable / in use tion as "free' le technolog	uidelines ","booked ies and co	applying to th ","in use" rresponding tes	ie testł	oed		
Com Subs	ponent or ystem	Testbed	s Directory S	Service							
Refi	nes/Replaces	PT-P-0	)3								

Id:	PT-DIR-S-002	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Architecture Deliverables	Ver:	2
Title	:	The Te Testbed	estbed Directed strength	tory Service s in RAWFIE ac	should provi cording to p	de access redefined	s to information filters	on on	all
Desc	ription:	The Te other R	stbed Direct AWFIE con	tory Service sinponents to be	hould provide able to filt	le the Wo	eb Service inte cess informati	erface on of	for the

	testbeds' registered in the RAWFIE database, according to specific filtering parameters (e.g. name, supported technologies)
Additional Info	
(comments):	
Component or	
Subsystem	Testbeds Directory Service
Refines/Replaces	

Id:	PT-DIR-S-003	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2	
Title	:	The Testbed Directory Service shall provide access to information about available resources (UxVs) belonging to the testbeds registered in RAWFIE								
Desc	cription:	The Te RAWF belongi	stbed Director IE compone ng to the diff	ory Service sha ents to be ab ferent testbeds	all provide the le to acces registered in	e Web Se s informa RAWFIE	ervice interface ation on the 2.	for ot resour	her ces	
Addi (com	itional Info ments):	Provided resources (UxVs) information includes: <ul> <li>name</li> <li>geographic location</li> <li>short description</li> <li>testbed to which the resource is associated</li> <li>type of resource (e.g. USV, UAV, etc)</li> <li>status (e.g. "free", "booked", "in use", "non operationa</li> <li>health status</li> </ul>								
Com Subs	ponent or system	Testbec	ls Directory S	Service						
Refi	nes/Replaces									

Id:	PT-DIR-S-004	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Architecture Deliverables	Ver:	2
Title:		The Terresource accordi	stbed Directo es (UxVs) ng to predefi	bry Service sho belonging to ned filters	buld provide the testbec	access to ls registe	information or red in RAW	navaila FIE, a	ble and

Description:	The Testbed Directory Service should provide the Web Service interface for other RAWFIE components to be able to filter and access information of the resources, according to specific filtering parameters (e.g. name, supported technologies)
Additional Info	
(comments):	
Component or	
Subsystem	Testbeds Directory Service
Refines/Replaces	

Id:	PT-DIR-S-005	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2	
Title	::	The Testbed Directory Service shoud provide the possibility to register new testbeds in the RAWFIE platform, as well as to unregister (delete) testbeds from the platform								
Desc	cription:	Each pa Platforr testbed The regi the regi CRUD resource	articipating to n. During i shall be prov staration ser stered data. I operations es	estbed shall be nitial registrat vided and store vice should all Basically, the 7 (CREATE, RI	registered i ion importa d in an appro ow for perio Festbed Direct EAD, UPDA	n order to nt details opriate tes dic or tes ctory Serv ATE, DEI	participate in needed to a tbed directory s tbed initiated u rice should pro- LETE) for test	RAWI ccess service updates vide ba tbeds a	FIE the of sic and	
Addi (com	tional Info ments):									
Com Subs	ponent or ystem	Testbed	s Directory	Service						
Refi	nes/Replaces	PT-P-0	)4							

Id:	PT-DIR-S-006	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Architecture Deliverables	Ver:	2	
Title	2:	Some basic query capabilities should be provided.								
Desc	cription:	Some b certain	asic query cara capabilities	apabilities show (testbed or/and	uld be provid l UxV resour	led. to fin rce specif	d resources that ic) that may ne	at prov ed for	ide an	

	experiment.
Additional Info	Need to define what exactly these capabilities could be for the testbed node and its various resources (i.e. CPU, RAM, Op. system, battery state, communication interfaces, sensor types, capabilities regarding resource controller, etc.)
(comments):	Need also to agree whether query capabilities would be available via an SQL query like language or via appropriate drop down menus or catalogues (the latter might be preferable for novice users but may limit the complexity of queries and consequently the granularity of searches).
Component or Subsystem	Testbeds Directory Service
Refines/Replaces	PT-A-016

Id:	PT-DIR-S-007	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2	
Title	::	The Testbed Directory Service shall provide the possibility to register new resources belonging to a specific testbed in the RAWFIE platform, as well as to unregister (delete) resources								
Desc	cription:	pdates of th bed Director EAD, UPDA	e registere y Service ATE, DEI	ed resources an e should prov LETE) for test	nd rela ide ba tbeds a	ted isic and				
Addi (com	itional Info ments):									
Com Subs	ponent or system	Testbec	ls Directory	Service						
Refi	nes/Replaces									

### 4.1.12 EDL Compiler and Validator

Id:	PT-CPV-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	DoW	Ver:	2		
Title:		A tool for translating EDL into user directives shall be provided									

Description:	The compilation and validation will be performed on top of the proposed EDL model that is based on a specific grammar. The compiler / validator will access the provided script and identify any errors that could jeopardize the execution of the experiment.
Additional Info	
(comments):	
Component or Subsystem	EDL Compiler & Validator
Refines/Replaces	PT-A-003

Id:	PT-CPV-002	Туре:	FUNC	Importance (priority):	HIGH	Source:	DoW	Ver:	2	
Title	:	An experimenter should have the opportunity to use a code generation engine								
Desc	cription:	When r final co	to errors are de to be uplo	present, the El baded in the Up	DL compiler «Vs.	and valid	ator should ge	nerate	the	
Addi (com	tional Info ments):	The conformat	de generatio that will be a	on module will adopted by the	transform t underlying n	he EDL o odes.	elements into	a spec	ific	
Com Subs	ponent or ystem	EDL Compiler & Validator								
Refi	nes/Replaces	PT-A-0	·-A-003							

Id:	PT-CPV-003	Туре:	FUNC	Importance (priority):	HIGH	Source:	DoW	Ver:	2
Title	le: Experiments defined via EDL shall be validated after their authoring								
Desc	cription:	Both te predefin availabi experin restricti	extually or wined set of rui ility of selement about ons/contradi	visually define- les (i.e. syntact octed resources t syntactic ctions.	d experimen tically, regar s) providing or sem	ts shall b ding spati feedbacl antic e	be validated ba al and/or spati- k to the auth- errors and	ased of otempo or of possi	n a oral the ble
Addi	itional Info								



(comments):	
Component or Subsystem	EDL Compiler & Validator
Refines/Replaces	PT-A-014

Id:	PT-CPV-004	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2	
Title	:	The compiler and validator should communicate with the authoring tool in order to transfer error indications and hints for solving them								
Desc	cription:	The con message experin will be	npiler and values to the energy to the energy and t	alidator will co experimenters and hints for s red in the front	ommunicate in order to securing the end.	with the fi provide correct ed	ront end layer help in edi- liting of the ex	to conv ting th perime	/ey neir ents	
Addi	itional Info									
(com	iments):									
Com	ponent or	EDL Co	ompiler & V	alidator						
Subs	ystem									
Refi	nes/Replaces									

# 4.1.13 Experiment Validation Service

Id:	PT-EXV-S-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	DoW	Ver:	2	
Title	RAWFIE shall provide a validator to constantly check exper during runtime							scenar	ios	
Desc	cription:	EVS witestbed. qualitat experin between control	ill validate in Cross exp ive character nent workflo n nodes will activities	f each experim periments vali ristics of an exp w, will retain s l be secured a	ent can effic dation will periment. For security and on s well as co	ciently be be perforr instance, qualitative ollision av	executed in the prmed accomp the EVS, base issues. Comm roidance and c	e selec panied ed on ea nunicat qualitat	ted by ach ion ive	
Addi (com	itional Info ments):	EVS pr of a set Handle function	EVS provides semantic validation for each experiment. It checks the fulfilm of a set of constraints defined by experts or Security Board. It can handle Handle security & safety issues e.g., collision avoidance, and other a functional (qualitative) aspects of each experiment. Efficient communicat							

	and control of the UxVs team will be performed in order to increase the performance of the system. It performs also cross experiment validation in order to help in maximising the performance of RAWFIE framework.
Component or Subsystem	Experiment Validation Service
Refines/Replaces	PT-L-001

Id:	PT-EXV-S-002	Туре:	FUNC	Importance (priority):	HIGH	Source:	DoW	Ver:	2	
Title	2:	The validation service should perform syntactic checking								
Desc	cription:	The ED the pro- identify experin	DL validation wided EDL any synta nent.	n service is res scripts. The actic errors th	ponsible for service will nat could jo	performi access t eopardize	ng syntactic an he provided s the executio	nalysis script a n of	on and the	
Addi (com	itional Info nments):	The service will syntactically check every script in terms of the EDL. Hints for correcting possible errors will be provided to the experimenters.								
Com Subs	ponent or system	Experiment Validation Service, EDL Authoring Tool								
Refi	nes/Replaces	PT-L-0	01							

Id:	PT-EXV-S-003	Туре:	FUNC	Importance (priority):	HIGH	Source:	DoW	Ver:	2
Title	2:	The validation service should perform semantic checking							
<b>Description:</b> The EDL validation service is resp the provided EDL scripts. The se identify any semantic errors also w infrastructure. It is capable of communication, spatio-temporal ma					ponsible for service will with the use f applying hanagement,	e performi access t e of data semantic sensing ar	ng semantic and he provided s stored in the u c checking f nd data manage	nalysis cript a inderly or not ement.	on and ing des
Addi (com	itional Info nments):	The service will semantically check every script in terms of the EDL. correcting possible errors will be provided to the experimenters.							for



Component or Subsystem	Experiment Validation Service
Refines/Replaces	PT-L-001

## 4.1.14 Users & Rights Service

Id:	PT-USR-S-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	DoW	Ver:	2	
Title	Title:         User login credentials checking shall be provided									
Desc	ription:	The login credentials of user shall be check, before the user may access any restricted services.								
Addi (com	tional Info ments):	Login via X.509 client certificate may also be possible.								
Com Subs	ponent or ystem	Users & Rights Service								
Refi	nes/Replaces	PT-GE	N-002							

Id:	PT-USR-S-002	Туре:	FUNC	Importance (priority):	HIGH	Source:	DoW	Ver:	2	
Title	:	The Users & Rights Service shall support various roles with different privileges at every level of access.								
Desc	cription:	The pla Every p exist: •	atform shall blatform user Experiment Admin Testbed Op	provide a set should be ass er erator	of different ign to a role.	roles wit At least	h predefined j the following i	orivileg roles sh	ges. nall	
Addi (com	itional Info nments):	Each of Definiti Each se proxy s	them provid ton of additionervice has to ervice may a	ling different a onal roles may o check if the llso be used tha	ccess rights t be possible. use has the tt restricts the	to the vari appropria e access to	ous platform so ate roles to ac the service.	ervices cess it.	. A	



Component or Subsystem	Users & Rights Service
Refines/Replaces	PT-GEN-002

Id:	PT-USR-S-003	Туре:	FUNC	Importance (priority):	LOW	Source:	Iteration1 Exp	Ver:	2		
Title	:	The Users & Rights Service may provide a proxy service for web application that do not check access rights.									
Desc	eription:	The pla page on	The platform may provide a proxy service that restricts the access to special web page only authorise users.								
Addi (com	tional Info ments):	Implem	Implementation specific, if this proxy needed.								
Component or Subsystem Users & Rights Service											
Refi	nes/Replaces										

### 4.1.15 Booking Service

Id:	PT-BOO-S-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2		
Title	:	Booking Service shall support reservations of resources at both user level and experiment level									
Description:		The Re two (2)	servation of l levels: The user l usually sho at this stage to be availa The experin validation of to specific of	Resources in R evel which is ould precede e e are not assig able for a user f ment level whi of an experime experiments of	AWFIE syst performed xperiments of ned to a speci- for the speci- ch is perform ent. This leve sequence of	by a pote lefinition. cific exper ied time f ned after ( el of reser	ected to be per ential experim The resources riment but are rame or during) auth vation assigns nts	formed enter a reserv guaran oring a resour	1 at and ved itee and ces		
Addi (com	tional Info ments):										



Component or Subsystem	Booking Service
Refines/Replaces	PT-B-001

Id:	PT-BOO-S-002	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2		
Title	:	User lev	User level booking shall be triggered by the Booking Tool via a REST API.								
Desc	cription:	The ma Tool in related	in way for a terface. This to a particula	user to reserve kind of reserv ar experiment	e resources w ation does no	vill be loca ot contain	te them via the any kind of in	e Book format	ing ion		
Addi (com	itional Info ments):										
Com Subs	ponent or ystem	Bookin	g Service								
Refi	nes/Replaces	PT-B-0	01								

Id:	PT-BOO-S-003	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2		
Title	2:	Experir a manua	Experiment level booking shall be triggered by the experimenter before issuing a manual or schedule launching of a validated experiment								
Desc	cription:	The res	The reservation of resources to specific experiments is achieved during authoring of an experiment and should precede the actual launching of the experiment								
Add (con	itional Info nments):										
Com Subs	ponent or system	Bookin	g Service								
Refi	nes/Replaces	PT-B-0	01, PT-L-00	2							



Id:	PT-BOO-S-004	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2		
Title	:	Experir booking	Experiment level booking shall support both immediate booking as well as booking at a future time								
Desc	ription:	Experir resourc	Experiment level booking shall allow an experimenter to define whether the resources are to be reserved directly or a a future timeslot.								
Addi (com	tional Info ments):	immedi	ate booking	will probably l	be initiated d	uring expe	eriment authori	ng.			
Com Subs	ponent or ystem	Booking Service									
Refi	nes/Replaces	PT-B-0	01								

Id:	PT-BOO-S-005	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2		
Title	:	Booking includir	Booking Service shall provide all the necessary methods to manage the bookings including addition, modification and cancellation/deletion operations								
Desc	ription:										
Additional Info (comments):											
Component or Subsystem		Booking	g Service								
Refi	nes/Replaces										

Id:	PT-BOO-S-006	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2
Title:		Bookin booking	g Service sh gs for a provi	all be able to ided booking re	compute ar equest	nd return	feedback on c	onflict	ing
Description:									
Add (con	itional Info ments):								



Component or Subsystem	Booking Service
Refines/Replaces	

Id:	PT-BOO-S-007	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2		
Title	:	Reserva availab	Reservation Data shall be persistent in order to survive service failures and be available by other services								
Desc	cription:	The re- persiste compor	servation in nce purpose nents (i.e. the	formation sho s as well as a launching ser	uld be back since it might vice for scher	ked in a ht be nee duled exp	relational dat ded by other eriments).	abase RAWI	for FIE		
Addi (com	itional Info ments):										
Com Subs	ponent or ystem	Booking Service									
Refi	nes/Replaces										

Id:	PT-BOO-S-008	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2		
Title	2:	Historio	Historical data retrieval for Bookings/Reservations shall be available on demand								
Description:		Persisted Booking information should be available for search and statistic purposes. Therefore booking information should maintain timestamps (for the start and end time of booking)									
Addi (com	itional Info nments):	Info s): Information should be available for both user level and experiment reservations							evel		
Com Subs	ponent or system	Bookin	g Service								
Refi	nes/Replaces										



Id:	PT-BOO-S-009	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2		
Title:		Booking functionality shall support reservation of resources involving multiple testbeds									
Description:		The booking module must allow for the purpose of a single experiment the possible reservation of resources from different physical testbeds if this is explicitly requested from an experimenter.									
Addi (com	itional Info nments):										
Com Subs	ponent or system	Bookin	g Service								
Refi	nes/Replaces	PT-B-0	03								

Id:	PT-BOO-S-010	Туре:	FUNC	Importanc (priority):	HIGH		Source:	Iteration1 Exp	Ver:	2
Title:		Bookin Reserva	g functionations reques	ality shall sts by end use	be able ers	to	correctly	handle sin	nultane	ous
Description:										
Additional Info (comments):										
Component or Subsystem		Bookin	g Service							
Refi	nes/Replaces	PT-B-0	03							

Id:	PT-BOO-S-011	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	2	
Title:		Notification mechanisms may be provided for experiments scheduled for execution in the future.								
Description:		A notification mechanism to remind an experimenter the date and the timeslot allocated for running his/her experiment on the RAWFIE infrastructure may also be envisaged to improve the user experience. The time of notification prior to the								



	experiment launch should be configurable.
Additional Info	
(comments):	
Component or Subsystem	Booking Service, Launching Service
Refines/Replaces	РТ-В-004

# 4.1.16 Launching Service

Id:	PT-LAU-S-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2	
Title	:	Launching Service shall support short-term or manual launching of an experiment initiated directly by an experimenter								
Description:		The Launching Service through a specific interface will give the opportunity to experimenters to execute in real time pre-defined and pre-approved experiments stored in the RAWFIE system.								
Addi (com	itional Info ments):									
Com Subs	ponent or system	Launch	ing Service							
Refi	nes/Replaces									

Id:	PT-LAU-S-002	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2
Title:Launching Service shall support long-term experiment initiated directly by an experimenter						or scheer	duled launchin	ng of	an
Description:		The Launching Service shall provide the ability to execute experiments at a future time based on the associated bookings/reservations. In order to do that the Launching Service may utilize an appropriate scheduler.							
Addi	tional Info								
(com	iments):								
Com	ponent or	Launch	ing Service						


Subsystem	
Refines/Replaces	

Id:	PT-LAU-S-003	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2	
Title	:	Each e ecosyst	Each executing experiment shall be uniquely identified within RAWFIE ecosystem							
<b>Description:</b> The Launching Service shall ensure that during launching a unique Idential associated with the experiment which can be used from any other comport service to reference the running experiment							lentifie: ponent	r is t or		
Addi (com	itional Info ments):									
Com Subs	ponent or system	Launch	ing Service							
Refi	nes/Replaces	PT-E-0	01							

Id:	PT-LAU-S-004	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2
Title	:	During launching it must be ensured that the experiment to be started has been validated based on spatio-temporal constraints							
<b>Description:</b> The Launching Service shall allow execution of experiments t validated based on spatial (usually imposed by an reservations/bookings) or temporal (usually based on information EDL script) constraint that may exist							riments that h by an exp formation pres	nave bo berimer ent in	een iter the
Addi (com	itional Info ments):								
Com Subs	ponent or ystem	Launch	ing Service,	Experiment Va	alidation Serv	vice			
Refi	nes/Replaces	PT-L-0	02						

Id:	PT-LAU-S-005	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2
Title	:	During launching it must be ensured that the experiment to be started belongs an authorized user of the RAWFIE platform							
Desc	<b>Description:</b> The Launching Service shall allow execution of experiments that have be issued by existing RAWFIE platform users. If e.g. a request is received for a u that is not active any more it should be discarded.							een Iser	
Addi (com	itional Info ments):								
Com Subs	ponent or ystem	Launch	ing Service,	Experiment Va	alidation Serv	vice			
Refi	nes/Replaces	PT-L-0	02						

Id:	PT-LAU-S-006	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2
Title	:	The Lau an expe	unching Serv riment	vice shall be ab	le to address	simultane	eous requests fo	or start	ing
Desc	ription:	The Lau an expe	unching Serv riment at a r	vice should be easonable time	able to hand and in a thre	le multipl ead safe m	e requests for nanner.	launch	ing
Addi (com	tional Info ments):								
Com Subs	ponent or ystem	Launch	ing Service						
Refi	nes/Replaces								

Id:	PT-LAU-S-007	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2
Title:         The Launching Service shall send an appropriate starting of an experiment					opriate m	essage upon s	success	ful	
Desc	ription:	<b>Detion:</b> The Launching Service shall provide an indication of successful experiment by publishing an appropriate message that contains the execution ID of experiment and possible additional information that may be needed by							tart the her



	services.
Additional Info	
(comments):	
Component or	I aunching Service
Subsystem	
Refines/Replaces	PT-E-001

Id:	PT-LAU-S-008	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2
Title	:	The Launching Service shall interact with other components or database service in order to retrieve information needed for deciding on launching an experiment							ces nt
<b>Description:</b> The Launching Service shall be able to directly int databases and possibly additional services or too Experiment Controller etc.) in order to figure out wh request should be issued.						ectly inte or tool out whe	ract with the ls (Validation ther an experin	RAWI Servi ment s	FIE ice, tart
Addi (com	itional Info ments):								
Com Subs	ponent or system	Launch	ing Service						
Refi	nes/Replaces								

Id:	PT-LAU-S-009	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title	:	Interact compor	ions of the nents should	e launching services and the services an	ervice with WFIE platfo	database orm bound	services and ary	/or ot	her
<b>Description:</b> The Launching Service should not be allowed to directly in components or services outside the RAWFIE platform. Direct c Testbed service i,e. should not be allowed.						directly inte m. Direct cal	ract w ling of	/ith f a	
Addi (com	itional Info nments):								
Com Subs	ponent or system	Launch	ing Service						



Refines/Replaces				
------------------	--	--	--	--

Id:	PT-LAU-S-010	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title	2:	Launch	ing service s	hall support re	quests for ex	periment of	cancellation		
Desc	cription:	Reques handled	ts for cancel l by the laun	ling an already	y running or	scheduled	d experiment s	hould	be
Add (con	itional Info nments):								
Com Subs	ponent or system	Launch	ing Service						
Refi	nes/Replaces								

Id:	PT-LAU-S-011	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	2
Title	2:	RAWF.	E platform on	shall provide	e means to	ensure f	fairness in ex	perime	nts
Desc	cription:	tion: RAWFIE platform shall provide mechanisms, either automated or involvin manual intervention (i.e. by an administrator) that will ensure fairness i experiments execution thus avoiding a resource being perpetually used by certain experiment.							ing in y a
Addi (com	itional Info nments):								
Com Subs	ponent or system	Launch	ing Service						
Refi	nes/Replaces	PT-L-0	07						

Id:	PT-LAU-S-012	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2

Title:	Launching service shall provide appropriate feedback to the requested entity regarding failures on fulfilling a request
Description:	If a request for starting or cancelling an experiment fails to be successfully processed by the Launching Service then an appropriate response should be returned indicating the reason of failure.
Additional Info (comments):	<ul> <li>Possible reason of failure may include (not exhaustive):</li> <li>Experiment already running</li> <li>Not existent experiment ID</li> <li>Experiment addressing not reserved resources</li> <li>Communication failure (inability to sent StartExperimentRequest</li> <li>Experiment with inconsistency regarding its initial execution time</li> </ul>
Component or Subsystem	Launching Service
Refines/Replaces	

Id:	PT-LAU-S-013	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title	:	Launch execution	ing service s	shall not alter of eriment	or modify an	y informa	tion related to	the act	ual
Desc	cription:	The pu unique the RA are not	rpose of lau Id capable o WFIE ecosy be handled b	inching service of being used to stem. Informatory by this component	e is to initia for identifyin tion related tent.	te an exp ng a "rum to the inte	periment and g ning" experime ernals of the e	enerate ent wit xperim	e a hin ent
Addi (com	itional Info ments):								
Com Subs	ponent or ystem	Launch	ing Service						
Refi	nes/Replaces								

## 4.1.17 Visualisation Engine

Id:	PT-VIS-E-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2
-----	--------------	-------	------	---------------------------	------	---------	------------------------------	------	---

Title:	The Visualization Engine shall retrieve from the message bus all runtime experiment information needed for visualizing the UxVs and/or any sensor measurments
Description:	During the experiment execution, the Visualisation Engine will be in charge of handling the communication with the Message Bus, in order to retrieve all the information (e.g. sensors measurements and position) that will be available during the experiment's execution
Additional Info (comments):	
Component or Subsystem	Visualisation Engine
Refines/Replaces	PT-L-005

Id:	PT-VIS-E-002	Туре:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2		
Title:		The Visualization Engine shall provide a GIS server capable of handling geographical layers (overlays)									
Desc	cription:	<ul> <li>The Visualisation Engine shall provide all server side functionalities to add and manipulate multiple geographic elements as overlays shall be possibile to add, organise and access to georeferenced eleusing one or more of the following technologies:</li> <li>Georeferenced information stored in the PostGIS database</li> <li>WMS layers from external providers</li> <li>WFS layers from external providers</li> <li>Shapefiles</li> </ul>						IS serv ne map ts (laye	ver) . It ers)		
Addi (con	itional Info nments):										
Com Subs	ponent or system	Visuali	sation Engin	e							
Refi	nes/Replaces										

Id:	PT-VIS-E-003	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Architecture Deliverables	Ver:	2

Title:	The Visualization Engine may allow cache of data for faster access to the available geographic layers
Description:	The GIS Server provided by the Visualisation Engine may provide caching functionality of geographic data, for faster loading time.
Additional Info (comments):	
Component or Subsystem	Visualisation Engine
Refines/Replaces	

Id:	PT-VIS-E-004	Type:	FUNC	Importance (priority):	HIGH	Source:	Architecture Deliverables	Ver:	2
Title:         The Visualization Engine shall provide the postusing historical data					ossibility	to reply ex	perime	ents	
Desc	cription:	The experimenter shall be able to choose, from the Visualisation experiment to be repeated. The request shall be handled by the Vis Engine, which retrieves information about the past experiment (includ maps and layers) directly from the database of through other M components.							the ion ited lier
Addi (com	itional Info ments):	By replaying any of the experiments, the user can gather or check data for an experiment at a convenient time after the experiment finished.							
Com Subs	ponent or system	Visuali	sation Engin	e					
Refi	nes/Replaces								

## 4.1.18 Experiment Controller

Id:	PT-EXP-C-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2		
Title:		Cancellation of running experiments should be possible									

Description:	Experiment controller should be able to receive from the experimenter instructions regarding the cancellation of an ongoing experiment. In the sequel, the experiment controller should forward these instructions to the resource controller
Additional Info (comments):	
Component or Subsystem	Experiment Controller
Refines/Replaces	

Id:	PT-EXP-C-002	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	2	
Title	:	RAWFIE platform shall allow experimenters to remotely navigate UxVs.								
Desc	cription:	RAWF through Either are tran with th controll	E experime a virtual rer the experime slated into a be build-in led units.	enters shall ha note controller enter directly c "global form" navigation sy	ave ability to provided by controls the of waypoint stem of the	the applic the applic UxV or the ts (a refer UxVs)	the unmanned cation's interfa he provided in ence scheme c and transmitte	l vehic ce. structio ompati ed to	ons ible the	
Addi (com	tional Info ments):	The virtual remote controller will act as proxy control unit communicating w the real control unit that lies on each testbed The exchanged messages should be designed in respect to open standa possibly using well know formats (i.e. JSON or XML).							/ith ards	
Com Subs	ponent or ystem	Experir	nent Control	ler						
Refi	nes/Replaces	PT-L-0	08							

Id:	PT-EXP-C-003	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title	:	The Experiment Controller shall support the execution of experiment involve multiple testbeds							hat
Desc	ription:	Experir geograp handle	Experiments written by users can involve resources that belong in diffe geographically dispersed locations. The experiment controller must be abl handle and coordinate all kinds of information exchange for all the diffe						



	testbeds participating in the experiment.
Additional Info	
(comments):	
Component or	
Subsystem	Experiment Controller
Refines/Replaces	

Id:	PT-EXP-C-004	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title:The Experiment Controller shall be able to support mult the same time in parallel						ple experimen	ts runn	ing	
Desc	cription:	As a multi-user environment multiple RAWFIE experiments can run in parallel in temporal dimension. The Experiment Controller must be able to smoothly support all the experiments that temporally coexist without degradation of service performance.							llel hly of
Addi (com	itional Info ments):								
Com Subs	ponent or system	Experiment Controller							
Refi	nes/Replaces								

Id:	PT-EXP-C-005	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title	2:	The Experiment Controller shall be able to analyse the whole experime and dispatch the appropriate parts to each responsible testbed facility							
<b>Description:</b> After receiving the validated EDL script, Experiment Contr process its content, identify the involved testbeds and ser responsible component (Resource Controller) only the infor- testbed.					ontroller must send to each formation rela	be able testbe ted to t	to d's his		
Add (con	itional Info nments):								
Com	ponent or	Experir	nent Control	ler					



Subsystem	
Refines/Replaces	

Id:	PT-EXP-C-006	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2	
Title	::	The Experiment Controller shall support receiving feedback at regular intervals from all testbed facilities about the progress of the experiment in this time interval								
<b>Description:</b> RAWFIE experiments can be expanded in different testbeds and Exp Controller as a coordination point must be aware in time about the pro- the experiment in all physical testbeds involved. Experiment Controller able to compose the whole picture of the experiment upon receiving f from the individual testbeds building a clear view of the overall status correctness of steps executed so far.						xperim rogress r must feedba is and	ent of be ack the			
Addi (com	itional Info ments):									
Com Subs	ponent or ystem	Experin	Experiment Controller							
Refi	nes/Replaces									

Id:	PT-EXP-C-007	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2	
Title	:	The Experiment Controller may be able to override the order of i described in the input script while the experiment is running								
Desc	cription:	Based have the may be	Based on the feedback from testbed facilities, the Experiment Controller may have the ability to override the future steps described in the input script. This may be done for safety or feasibility reasons.							
Addi (com	tional Info ments):	This ca always	This can of override should be allowed only after special authorization an always respecting the constraints of each testbed facility.							
Com Subs	ponent or ystem	Experir	Experiment Controller							
Refi	nes/Replaces									

Id:	PT-EXP-C-008	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title:The Experiment Controller shall be able to controller (Experiment Monitoring Tool) giving the experiment workflow as a whole						ntinuousl xperiment	y feed the fror er a clear vie	it-end w of	tier the
Desc	Experiment Controller is the responsible component for composing the while picture of the experiment and its progress compared to the aimed target and n send this information in front-end components and user interfaces through while the user interacts, giving the experimenter the ability to have a clear assessing about the experiment progress.							he wh and m igh wh ssessm	ole iust ich ent
Addi (com	tional Info ments):								
Com Subs	ponent or ystem	Experir	xperiment Controller						
Refi	nes/Replaces	PT-L-0	04						

Id:	PT-EXP-C-009	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title	:	The Exervery c	The Experiment Controller shall send distinct error and warning messages every case the experiment's state diverges from the aimed target						
Desc	cription:	End use the exp notifica	End users must have a clear view of the errors and warnings that occur during he experiment execution and Experiment Controller must provide all error notifications that occurred in its domain.						
Addi (com	itional Info iments):								
Com Subs	ponent or ystem	Experir	Experiment Controller						
Refi	nes/Replaces								

#### 4.1.19 Data Analysis Engine

The Data Analysis Engine is the intermediary between the analysis tool and spark. Spark is a distributed compute platform that can effectively factor out computations such as BLAS



operations. We utilize the Spark jobserver API to do most of the communication between our tool and the compute cluster.

Id:	PT-DAA-S -001	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2					
Title	:	Analysi	s engine wil	l support accer	oting of analy	vsis jobs								
<b>Description:</b> The Data Analysis Engine will provide a schema based approach where accept analytics jobs. Will also provide access to spark transparently.						ere it v	vill							
Addi (com	tional Info ments):	<ul> <li>Data analytical software will deliver a set of analytical functionalities such as:</li> <li>outlier detection,</li> <li>distribution shift detection,</li> <li>classification.</li> <li>Dimensionality reduction</li> <li>The end user will also be able to deploy custom jobs by posting a model ['jar'] to the jobserver.</li> </ul>						] to						
Com Subs	ponent or ystem	Data A	Data Analysis											
Refi	nes/Replaces	PT-E-0	04, PT-E-003	5				PT-E-004, PT-E-005						

Id:	PT-DAA-S -002	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2
Title:         Analysis engine will support compiling analysis						s jobs		•	
<ul> <li>The Analysis Engine supports posting models ['jar'] and parar ['job'].</li> <li>For obvious reasons a model should be decoupled from it's parar because we might want to post models of the same type with dif parameters and/or working on different metrics.</li> <li>A job is either a streaming job or a batch job. A streaming job has r of life, while a batch job does.</li> </ul>					aramet aramet differ as no e	ters ters ent end			
Addi (con	itional Info nments):								
Com Subs	ponent or system	Data A	nalysis						



Refines/Replaces	PT-E-005

## 4.1.20 System Monitoring Service

Id:	PT-SYM-S-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	Consortium	Ver:	2	
Title	:	RAWFIE middle tier shall include a module to monitor the performance of the middle tier components.								
<b>Description:</b> This module will check the performance Indicate software modules will perform at op					formance of ors (KPI) ar optimum leve	the mid nd this wa els.	dle tier compo y ensure that a	onents all criti	by cal	
Addi (com	tional Info ments):	Indicators could be: CPU load, free disc space, availability of system services (SSH, web server, etc.), availability and response time of the web services and databases servers etc.								
Com Subs	ponent or ystem	System	ystem Monitoring Service							
Refi	nes/Replaces	PT-GEI	<b>T</b> -GEN-004							

Id:	PT-SYM-S-002	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title	2:	RAWE	RAWFIE Testbeds and UxVs statuses should be monitored						
Desc	cription:	This module will collect the availability information of testbeds and UxVs.							
Addi (com	itional Info ments):	Testbec availabi (e.g. via Testbec as offlin	availability by their own and have to send them to the monitoring compone (e.g. via the message bus). Testbeds and UxVs that did not sent status updates for a long time are consider as offline.						neir ent red
Com Subs	ponent or system	System	System Monitoring Service						
Refi	nes/Replaces	N/A	N/A						



Id:	PT-SYM-S-003	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title	:	RAWF. platfror	AWFIE system administrators should be informed if critical, for the RAV latfrom operation, services are down						FIE
Desc	ription:	Emails critical	Emails should be sent to the system administrators if the monitoring consider critical components as down.						
Addi (com	tional Info ments):								
Com Subs	ponent or ystem	System	Monitoring	Service					
Refi	nes/Replaces	(PT-NF	5-007)						

Id:	PT-SYM-S-004	Туре:	FUNC	Importance (priority):	LOW	Source:	Iteration1 Exp	Ver:	2
Title:         User may register for notifications if certain components are down									
Desc	eription:	Emails should be sent to the users if the monitoring considers critical components as down.							
Addi (com	tional Info ments):								
Com Subs	ponent or ystem	System	Monitoring	Service					
Refi	nes/Replaces	(PT-NF	-007)						

Id:	PT-SYM-S-005	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2	
Title:		Notifications about planned downtimes								
Description:		Emails be dow:	should be se n.	ent to the intere	ested users if	some con	mponents are p	olannec	l to	
Addi (com	itional Info ments):									



Component or Subsystem	System Monitoring Service
Refines/Replaces	(PT-NF-007)

# 4.1.21 Accounting Service

Id:	PT-ACC-S-001	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	DoW	Ver:	2
Title	:	The ac regardin	The accounting service should be capable to accept different cost models regarding RAWFIE usage on a per service basis						
Desc	cription:	The main role of the accounting service will be to provide an effective co model for charging users of the platform based on the type of experiment and th services used. Different cost models should be supported and be configurable is terms of parameters.						ost the in	
Addi (com	tional Info ments):	In the early days of the federation and while the RAWFIE platform is in the phase of development and evaluation virtual credit units may be used to enable policy of fair resource sharing among users while after the EU funding period the accounting system can be used for applying a cost model viable for commercia use based on quantification of all costs involved in setting up, maintaining developing and managing the different facilities that are part of the federation							the le a the cial ng,
Com Subs	ponent or ystem	Accoun	ting Service						
Refi	nes/Replaces	PT-B-0	07						

Id:	PT-ACC-S-002	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	DoW	Ver:	2
Title	:	The acc the plat	he accounting service should be capable to gather statistics regarding usage of the platform by experimenters.						
Desc	cription:	The accounting service should be available from the early days of RAWF federation and ensure that all information pertaining to the use of the platfor and its services by potential experimenters is available.						FIE >rm	
Addi (com	itional Info nments):								
Com	ponent or	Accoun	ting Service						



Subsystem	
Refines/Replaces	РТ-В-007

Id:	PT-ACC-S-003	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	DoW	Ver:	2
Title	:	The RAWFIE platform should record information related to time and type of access for a service by a user.							
Desc	ription:	Informa exploite	ntion on whe ed later on fo	en and what ty r determining	pe of servic the platform	e each us accountin	er interacts wi g issues.	th can	be
Addi (com	tional Info ments):								
Com Subs	ponent or ystem	Accoun	ting Service						
Refi	nes/Replaces	PT-B-0	07						

Id:	PT-ACC-S-004	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2
Title	:	The cos execute	st model used d by a user o	d may take into of the platform.	o consideratio	on the ove	erall time of ex	perime	ents
Desc	cription:								
Addi (con	tional Info ments):								
Com Subs	ponent or ystem	Accoun	ting Service						
Refi	nes/Replaces	PT-B-0	07						



Id:	PT-ACC-S-005	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2
Title		The acc of the e	counting serv	vice may suppo (industrial, res	rt different t earch, univer	ypes of ch rsity etc.)	arging based o	n the ty	ype
Desc	cription:								
Addi (com	tional Info ments):								
Com Subs	ponent or ystem	Accoun	ting Service						
Refi	nes/Replaces	PT-B-0	07						

Id:	PT-ACC-S-006	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2	
Title	:	The acc usage o	The accounting service may support predefined types of memberships regardin usage of the platform that may depend on various types of parameters							
Desc	ription:									
Addi (com	tional Info iments):									
Com Subs	ponent or ystem	Accoun	iting Service							
Refi	nes/Replaces	PT-B-0	07							

Id:	PT-ACC-S-007	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2
Title:		The acc may be	counting serv	ice should be a l in the RAWF	able to handl IE platform o	e the addi luring tim	tion of new ser e.	vices t	hat

Description:	The accounting service must be able to update the applied cost model for services that possibly may be added after its initial deployment
Additional Info	
(comments):	
Component or	A second in a Gammin
Subsystem	Accounting Service
Refines/Replaces	РТ-В-007

## 4.2 Testbed Requirements

Testbed requirements include all the requirements pertaining the testbed facility components. The testbed components are mainly used for interconnecting with the RAWFIE server platform and for managing the UxV resources.

#### 4.2.1 General

Id:	TB-GEN-R-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2			
Title:		Each U capabil	Each UxV Testbed should provide a Slice Interface for federating their capabilities/resources to the experimenter.									
Desc	cription:	In accominima and be resourc The slice	ordance with 1 interface to longing to d es to their ov ce interface i	the general enable the fea different admi wners. s used to create	SFA concep deration of te nistrators, w e and control	ot each te estbeds wi hile gran slices.	stbed should th different tec ting the contr	provide hnolog ol of	e a gies the			
Add	itional Info											
(con	nments):											
Com	ponent or											
Subs	system											
Refi	nes/Replaces											

Id:	TB-GEN-R-002	Туре:	ENV	Importance (priority):	HIGH	Source:	Other	Ver:	1
-----	--------------	-------	-----	---------------------------	------	---------	-------	------	---

Title:	Each Testbed should provide the exact boundaries within which its UxVs can operate
Description:	The spatial boundary where UxVs can operate within a testbed should be predefined a priori. Any attempt of a UxV to move outside this boundary should be prohibited. Also requests by ground components attempting to breach the operating boundary should be rejected.
Additional Info	
(comments):	
Component or	
Subsystem	
Refines/Replaces	

Id:	TB-GEN-R-003	Туре:	FUNC	Importance (priority):	HIGH	Source:	Other	Ver:	1	
Title	:	Testbed areas should at least be able to host/operate multiple UxVs of one or more types								
Desc	cription:	Testbec hosting Additio enough	l areas shoul at least one nally, the e	ld provide eith e of three type xtend/size of t	er indoor ar es of unman the hosting	nd/or outd ned vehic testbed ar	oor facilities o eles (UAV,US) rea should be	capable V, UG suffici	of V). ent	
Addi (com	tional Info ments):									
Com Subs	ponent or ystem	N/A								
Refi	nes/Replaces									

Id:	TB-GEN-R-004	Туре:	ENV	Importance (priority):	HIGH	Source:	Other	Ver:	2
Title	2:	Testbed areas environment should be closely monitored							
Description:		Testbed environ the indo a much	l areas that li ment depend oor testbeds a more contro	ve demonstrat ling on the ext are physically lled environme	ion will take end of the o smaller than ent.	place, shoul utdoor and i the outdoor	d provide a ndoor space ones and be	control . Name nefit fr	led ely, om

Additional Info	
(comments):	
Component or	
Subsystem	N/A
•	
Refines/Replaces	TB-G-002

Id:	TB-GEN-R-005	Туре:	ENV	Importance (priority):	HIGH	Source:	Other	Ver:	2	
Title	:	Indoor spaces of a testbed should provide a controlled indoor environment								
Desc	cription:	Indoor in outd increase	spaces shoul loor space), e reproducibi	d be used in or all communic ility of results	der to evaluate to evaluate to evaluate to evaluate to evaluate to evaluate the evaluate the evaluate to evaluate the eva	ate, at earl other secu	ier stage (befo ırity checks.	re go-l This v	ive vill	
Addi	tional Info									
(com	nments):									
Com Subs	ponent or ystem	N/A								
Refi	nes/Replaces	TB-G-0	002							

Id:	TB-GEN-R-006	Туре:	SUPP	Importance (priority):	HIGH	Source:	Other	Ver:	2
Title	:	Testebed facility areas should comprise storing spaces and be able to receive inspect and assemble and/or fix UxVs							
Desc	eription:	All con mainter	nforts should nance, inspec	l be provided, ctions and mon	in terms of b itor.	oig, storing	g and spaces f	for Ux	√'s
Addi (con	tional Info ments):								
Com Subs	ponent or ystem	N/A							
Refi	nes/Replaces	TB-G-0	002						

Id:	TB-GEN-R-007	Туре:	SEC	Importance (priority):	HIGH	Source:	Other	Ver:	2		
Title	*	Testbed facilities should provide emergency services in an ext						y event	-		
Desc	cription:	Each te staff for	Each testbed facility should have a security/emergency plan and relevant trained staff for common extraordinary events, such as fire, crash.								
Addi (com	itional Info ments):										
Com Subs	ponent or ystem	N/A									
Refi	nes/Replaces	TB-G-0	002								

Id:	TB-GEN-R-008	Туре:	ENV	Importance (priority):	HIGH	Source:	Other	Ver:	2	
Title	:	Testbed areas should provide proper facilities and equipment								
Desc	cription:	Facility may inc	v area should clude (depend radar, cameras, antennas, receivers, optical track video servic	I have appropriding on the cas king, ces.	riate ground se):	– based and	l mobile equip	ment t	hat	
Addi (con	itional Info ments):									
Com Subs	ponent or system	N/A								
Refi	nes/Replaces	TB-G-(	002							



Id:	TB-GEN-R-009	Туре:	ENV	Importance (priority):	HIGH	Source:	Other	Ver:	1
Title	:	Testbed must provide dedicated computational resources							
Desc	cription:	Testbec very-hig system.	l must provi gh-bit-rate d	ide either a co ligital subscrib	ommitted PC ber line, abl	Cs and/or e to host	Virtual Mach and support	ines, w RAWI	vith FIE
Addi	tional Info								
(con	nments):								
Com	ponent or								
Subs	ystem	IN/A							
Refi	nes/Replaces								

Id:	TB-GEN-R-010	Туре:	ОТН	Importance (priority):	HIGH	Source:	Other	Ver:	1
Title		Testbeds should be supported by on-site personnel							
Desc	cription:	During provide chargin	testbed der d. Assigned g, maintenar	monstrations t personnel is nee and upgrade	he physical important fo es	presents or technica	of personnel al support, Ux	must V batt	be ery
Addi (com	itional Info ments):								
Com Subs	ponent or system	N/A							
Refi	nes/Replaces								

Id:	TB-GEN-R-011	Туре:	SEC	Importance (priority):	HIGH	Source:	Other	Ver:	1
Title:		Testbed	ls should con	form to all leg	al regulation	s and rest	rictions		

Description:	Testbeds areas should adhere and follow all legal restrictions that are applicable, according to specific laws and regulations, at local national and EU level that can be applied
Additional Info	
(comments):	
Component or	NT/A
Subsystem	N/A
Refines/Replaces	TB-NF-G-005

# 4.2.2 Monitoring Manager

Id:	TB-MOM-001	Туре:	DATA	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title	:	The Monitoring Manager component should be able to provide information about the capabilities of each resource node.							
Desc	cription:	Testbec availab load, fr	l's monitorin le resources ee RAM, bit	ng component s (i.e. for each error rate, etc.	should check node) in the	periodically facility lik	the current sta e battery lifeti	atus of ime, C	the PU
Addi (com	tional Info ments):								
Com Subs	ponent or ystem	Monito	ring Manage	r					
Refi	nes/Replaces	TB-G-(	)04, TB-G-00	06					

Id:	TB-MOM-002	Туре:	DATA	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title:		The Metersteed	onitoring Ma facilities	anager compo	nent should	collect and	report current	status	of
Description:		Testbec facilitie	l's monitorin es like weathe	ng component s er conditions, r	should check network conn	periodically ections avai	the status of t lable, etc.	he testl	bed

Additional Info	
(comments):	
Component or Subsystem	Monitoring Manager
Refines/Replaces	TB-G-001

Id:	TB-MOM-003	Туре:	DATA	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2	
Title	:	The Monitoring Manager component should store periodically all testbed information								
Desc	ription:	Testbed monitoring manager should collect and store the status of the testbe characteristics and the devices in a data log file, with a specific timestamp.							oed	
Addi (com	tional Info ments):	This ha System	s to be feasil Monitoring	ole as in some s Service, will n	specific cases ot exist.	s communica	ation with other	r tiers,	i.e.	
Com Subs	ponent or ystem	Monito	Aonitoring Manager							
Refi	nes/Replaces	TB-G-(	)03							

Id:	TB-MOM-004	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2	
Title	:	Testbed monitoring manager should be able to transmit the current status to the System Monitoring Service.								
Desc	eription:	Monito update	Monitoring Manager component should have the role of a special plugin which wi update the System Monitoring Service of the current status.							
Addi (con	tional Info ments):									
Com Subs	ponent or ystem	Monito	ring Manage	er						
Refi	nes/Replaces	TB-G-(	)03							



### 4.2.3 Network Controller

Id:	TB-NEC-001	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	1	
Title		The RAWFIE communication resources shall be managed in order to offer seamless connectivity in the normal operations of the system.								
<b>Description:</b> The RAWFIE Communication Manager will manage and optimize the use allocation of the communication resources. This is the case in particular respect to the communication link and its associated quality of service as we the possible switching between the two available communication links.							e use a cular w as well	and vith l as		
Addi (com	itional Info ments):	This pr commu	rocess is do nication met	one in real-tin rics. It is done	me on the in conjunction	basis of on with the	the monitorir e Resource Co	ng of ntroller	the :	
Com Subs	ponent or ystem	Networ	Jetwork Controller, Resource Controller							
Refi	nes/Replaces	PT-L-0	'T-L-009, TB-G-008							

Id:	TB-NEC-002	Type:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	1	
Title		Provision of network communication resource								
Desc	eription:	Provision network communication with Resource Controller								
Addi (com	tional Info ments):									
Component or Subsystem Network Controller, Resource				, Resource Cor	ntroller					
Refi	nes/Replaces									

Id:	TB-NEC-003	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	1		
Title	:	Alternative communication system									
Description:		Enable switching between available network technologies									

Additional Info (comments):	This feature should be offered on a per connected entity basis (e.g. a UxV), depending on the communication quality between this entity and the Testbed.
Component or Subsystem	Network Controller, Resource Controller
Refines/Replaces	TB-R-013

Id:	TB-NEC-004	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	1	
Title	2:	Manage	Management of the communication system							
<b>Description:</b> The UxV shall regularly check the communication status to de disconnection, defective link or degradation of the quality of service							etect a	any		
Addi (com	itional Info nments):	This feature is bilateral and it shall be present on both sides of communication: the communicating entity (e.g. a UxV) and the Testbed. This particularly useful when the UxVs are moving in an environment with obstact between them and the other components.							the s is cles	
Com Subs	ponent or system	Network Controller, Resource Controller								
Refi	nes/Replaces	TB-NF	-G-006							

Id:	TB-NEC-005	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	1
Title	:	Time co	onstraint veri	1			ı		
Desc	cription:	The network controller shall verify during the execution of the experiment that the time constraints specified on the exchanged data for the different types of UxVs are met. Whenever such time constraint is not met, this event shall be notified to the Experiment Controller and the resource controller, so that they can take the appropriate measures							
Addi (com	tional Info ments):	Measures include relaxing the constraint, switching to other resources ( alternative communication system), re-balancing the existing resources, stopp the experiment, etc.							e.g. ing
Com Subs	ponent or ystem	Networ	k Controller,	, Experiment c	ontroller, Re	source Co	ntroller		



Refines/Replaces	

#### 4.2.4 Resource Controller

Id:	TB-REC-001	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	2		
Title:		RAWFIE platform shall support a semi-autonomously way of navigation of the UxVs									
Desc	cription:	Experimenters provide details about the mission that UxVs will execute as well as comprehensive information about the algorithms to be used to process this task. RAWFIE undertakes the evaluation of all the employed elements and in each time step the system assesses the validity of the decisions of the involved algorithms. The internal control mechanism alters the trajectory of the units so as to ensure both, the vehicle's safety and the success of the mission. At each time step next optimum/appropriate waypoint for each UxV is transmitted to it.									
Addi (com	itional Info ments):	Real time tracking may be restricted by the communication technology of UxV data transmission. Cloud Technology may be not fast enough for real tracking.							the me		
Com Subs	ponent or system	Resource Controller									
Refi	nes/Replaces	PT-L-0	PT-L-009, TB-G-008								

Id:	TB-REC-002	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2		
Title	:	RAWF	IE platform s	should be able	to activate th	activate the "Emergency Scenario"					
Description:		The "R intende followi emerge	Remote Cont d and addition ng condition ncy scenario The compor- time steps The compor- localization	rol" compone onally, guarant ns occurs, a nent does not r nent receives issues	nt ensures t eees the safet utomatically eceive any f feedback fro	hat the sy ty of the e , the co eedback from the un	ystem is perfo equipment. If o omponent action rom the units f nits which repo	orming one of vates or seve ort sev	as the an eral		
Addi (com	tional Info ments):										



Component or Subsystem	Resource Controller
Refines/Replaces	PT-L-009, TB-G-008

Id:	TB-REC-003	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2	
Title	:	The Resource Controller shall receive location messages from the vehicles at regular intervals								
Desc	cription:	The Resource Controller shall be able to receive communication messages with the actual position's coordinates at regular intervals and in near real-time constraints. The Resource Controller shall be able to utilize this information for position variation estimation compared to the planned path, trajectory optimization, obstacles avoidance and identification of possible safety violations.								
Addi (com	itional Info ments):									
Com Subs	ponent or system	Resource	ce Controller							
Refi	nes/Replaces	TB-G-0	005, TB-G-00	03						

Id:	TB-REC-004	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2	
Title	2:	The Resource Controller shall transmit the next location for the current experiment to the vehicles								
Desc	cription:	The Real UxV ta Experir previou obstacle algorith	source Contr king into ac nent Control s, for the es es and the im.	oller shall be a count the exp ller and the act stimation of the system dynam	able to transmerimenter's tual position tual next point tual will be	nit the nex instruction received t the mod used as	t navigation points as received from UxVs. A lel of UxVs, r inputs in the	oint of from Above avigat plann	the the the ion ing	
Addi (com	itional Info nments):									
Com Subs	ponent or system	Resourc	ce Controller							



Refines/Replaces	TB-G-008

Id:	TB-REC-005	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2	
Title	:	The Resource Controller shall be able to plan the next location that will be transmitted in the vehicle taking into account the locations of all UxVs that are active in that testbed								
Desc	ription:	Taking into account that the Resource Controller shall be able to receive the actual locations of all UxVs at regular intervals and in near real-time constraints the component shall be able to utilize this information in the next steps for mission optimization and UxVs collision avoidance.								
Addi (com	tional Info ments):									
Com Subs	ponent or ystem	Resourc	ce Controller							
Refi	nes/Replaces									

Id:	TB-REC-005	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2	
Title	:	For the experiment accomplishment the Resource Controller shall operate in close coordination with the Experiment Controller								
<b>Description:</b> Resource Controller is responsible for transformed the Experiment Controller which is responsible for transformed the Experiment. Resource controller shall instructions after processing performed corrective instructions in case needed for experiment as a whole.					for trajector all report the s responsible shall be brmed in the eded for the	y monitor experime for the p able to Experime successfu	ing and optimi nt execution p rogress estimat receive exper ent Controller l accomplishm	zation rogress ion of rimente as well ent of	for the the r's as the	
Addi (com	itional Info ments):									
Com Subs	ponent or ystem	Resource	ce Controller							
Refi	nes/Replaces	TB-I-00	)1, TB-G-00	5						

# 4.2.5 Testbed Proxy

Id:	TB-PRO-001	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	2	
Title	:	Testbed proxy should act as a reverse proxy								
Desc	cription:	Testbec forward relevan Therefo	<ul> <li>Testbed Proxy represents a gateway between the middle and the testbed tie forwards the messages from the components that belong to middle tier to relevant components of testbed tier.</li> <li>Therefore Testbed Proxy</li> <li>accepts requests from middle tier and forwards only valid request testbed components.passes</li> <li>replies from testebed components back to middle and frontend tier</li> </ul>						to	
Addi (com	tional Info ments):									
Com Subs	ponent or ystem	Testbec	l Proxy							
Refi	nes/Replaces									

Id:	TB-PRO-002	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2	
Title	:	Testbed proxy contains Inner and Outer Firewall								
Desc	cription:	Testbec reach t firewall resultin the tes Monito	l Proxy filter he Resource is ensure that g network to stbed compo- ring Manage separates se denies inbou denies outbo filters incom Reverse Pro can deny ou	rs all requests, e Controller a t no external mo- pology provide onents of Re r. Testbed prov- rver zone from und connection ound connection ning network to oxy itbound connect	so that only nd the Test etwork traffic es a demilitate source Cor source Cor by DMZ as except from on from back raffic and all etion from Re	y (mostly) bed Mana c reaches rized zone ntroller, 7 m Reverse end servic ows only everse Pro	) harmless req ager. Two pac the real web so (DMZ) contai Festbed Mana Proxy res HTTP port acc xy	uests v eket fi erver. T ining o ager, a	vill lter The nly and the	
Addi (com	tional Info ments):									



Component or Subsystem	Testbed Proxy
Refines/Replaces	

# 4.2.6 Testbed Manager

Id:	TB-MAN-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	Consortium	Ver:	2	
Title	:	Testbed Manager shall support permanent storage of all testbed attributes and resources attributes that belong to testbed								
<b>Description:</b> Testbed Manager will be connected to a local database responsible for storage of all the items that exist within the boundaries of each testbed must include testbed and resources description, utilization of reso experiments running at testbed and logging of past activities.							e for bed. T resourc	the his es,		
Addi	tional Info									
(com	iments):									
Com Subs	ponent or ystem	Testbed	Manager							
Refi	nes/Replaces	TB-D-0	01							

Id:	TB-MAN-002	Туре:	FUNC	Importance (priority):	HIGH	Source:	Consortium	Ver:	2
Title	:	Testbed resource	Manager : e node	shall provide	information	about tl	ne capabilities	of ea	ach
Description:		Testbed capabili •	Manager ities. Such in HW charac Communic networking Sensing caj measureme	has to provid formation for teristics (CPU ation capabil interfaces, sof pabilities ent resource typ	de a compl UxV nodes n architecture ities (i.e. tware define	ete set of nay includ and speed supporte d radio)	of each resou le: l, RAM). d network s	rce no	de ds,
Addi (com	tional Info ments):								



Component or Subsystem	Testbed Manager
Refines/Replaces	TB-G-004

Id:	TB-MAN-003	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2	
Title	:	Testbed Manager shall check periodically the status of all other services running at testbed level								
Desc	cription:	Testbec that bel abnorm transmi represe	l Manager m ong to testb al operation tted to Sy ntation of the	nust be aware ed software ar or non-response rstem Monitor e status of all T	of the curre ad inform the nsive compo- ring Service 'estbed comp	nt status e platform onents. Th e making onents.	of all other co n in case of de nis information g available t	mpone tection 1 must he ex	nts of be act	
Addi (com	tional Info ments):									
Com Subs	ponent or ystem	Testbec	l Manager							
Refi	nes/Replaces									

Id:	TB-MAN-004	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2	
Title	:	Testbed Manager shall contain a registration log for all the experiments executed in the testbed								
<b>Description:</b> Testbed Manager maintains a history log in the local database experiments that conducted in the testbed giving to the testbed ability to have a clear picture of its previous utilization.						l database wi ne testbed ope	th all rators	the the		
Addi (com	tional Info ments):									
Com Subs	ponent or ystem	Testbec	l Manager							
Refi	nes/Replaces	TB-D-0	)02							

Id:	TB-MAN-005	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2	
Title	:	Testbed Manager shall be periodically informed about the status of all running experiments in the testbed								
Desc	cription:	After a notifica etc).	n experimer tions about	nt start, Testbe the status of th	ed Manager ne experimen	must be ant (ongoin	able to receive ag, completed,	e perio cancel	dic led	
Add (con	itional Info ments):	This no with Ux	This notification will be available from Resource Controller after communication with UxVs participating in the experiment.							
Com Subs	ponent or system	Testbed	l Manager							
Refi	nes/Replaces									

Id:	TB-MAN-006	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Iteration1 Exp	Ver:	2
Title	:	Testbed Manager shall store configuration parameters for the UxVs in relevant testbed							the
Desc	ription:	tion: Configuration parameters of UxVs may include communication interfiprograms and algorithms running in CPU, calibration and configuration sensors. These parameters may need a proper adjustment prior making U available to participate in RAWFIE experiments. A history log of every configuration may be stored in local database as well.							ces, of Vs xVs xV
Addi (com	tional Info ments):								
Com Subs	ponent or ystem	Testbed	l Manager						
Refi	nes/Replaces	TB-G-0	004						

Id:	TB-MAN-007	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title:		Testbec between	l Manager s n testbed ope	hall implement prators and mac	t a user inter thines	erface to	support the in	teractio	ons

Description:	A graphical user interface must be able to represent all the information about testbed attributes and its resources, ongoing experiments and logging activities of past experiments, testbed services running, resources configurations and any other information related to testbed administration. The information stored in the local database shall be displayed to the testbed operator through this interface.
Additional Info (comments):	
Subsystem	Testbed Manager
Refines/Replaces	

Id:	TB-MAN-008	Туре:	DATA	Importance (priority):	HIGH	Source:	Consortium	Ver:	2
Title	:	Testbed Manager shall be capable to handle temporary interruption of communication and store data locally in case of transmission failure							
Desc	ription:	Capabil loss bet of data	ity of enabl ween Testbe as soon as th	ing local data and the rest link is establ	storage in c of RAWFIE ished again.	case of tra	nsmission fail (ture) and retra	ure (L nsmiss	ink ion
Addi (com	tional Info ments):								
Com Subs	ponent or ystem	Testbec	l Manager						
Refi	nes/Replaces	TB-D-0	001						

Id:	TB-MAN-009	Туре:	DATA	Importance (priority):	LOW	Source:	Consortium	Ver:	2				
Title:		Testbec	Testbed Manager may provide statistical data/information about testbed										
		operation											
Description:		Statistical data such as: number of experiments; experiments duration; number of											
	-	UxV nodes used; Testbed time alive; etc.											
Add	itional Info												
(con	nments):												



Component or Subsystem	Testbed Manager
Refines/Replaces	TB-D-002

### 4.3 UxV Requirements

This subsection includes requirements related to UxVs and the expected functionality. In order to participate in RAWFIE experiments the UxV should implement a minimum level of common functionality irrespective of their type (UGV, UAV, USV) in terms of communication capabilities, on-board processing capabilities end localization.

4.3.1 General

Id:	TB-UVG-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	1		
Title	2:	Compli	ance of UxV	to RAWFIE s	pecification	and interfa	aces		<u>,                                     </u>		
Desc	cription:	To be able to operate in a RAWFIE Tesbed, a RAWFIE UxV interacts with the other Testbed entities (proxy, controllers, other UxV's). As such the UxV shall conform to the RAWFIE global architecture and conceptual components defined in D4.2.									
		The Uz vehicle it appea robot o time rei	The UxV Node component provides an abstraction layer to the unmanned vehicle systems (such as ROS and other proprietary operating systems) to make it appearing as a RAWFIE compliant component. It provides interfaces to the robot operation resources such as setting the robot waypoints and speed or real-time remote control								
Additional Info (comments):		time remote control. The UxV shall for example provide a minimum set of capabilities to the RAWFIE system. The minimum set of features is a subset of the following items: Processing capabilities (type of processors, number of cores, speed); Size and dimensions; Weight; Payload; Battery; Number and type or sensors; Number and type of integrated network components and supported communication interfaces; Minimum and maximum autonomy of the device; Auto-return capability (return to the base station automatically); Ability of the vehicle to operate as an access point; (Remote) Control interface; Over-the-air programming capabilities; Provision of collision avoidance mechanism; Compatibility with Apache Kafka architecture; Data storage of the vehicle; Support of "safe mode" operation; Localization capabilities (e.g., GNSS); Ability to operate in indoor/outdoor/mixed environments; Compliance with standards,									
Component or Subsystem     UxV proxy and adapter											



Refines/Replaces	

#### 4.3.2 UxV Node

The UxV Node provides an interface to the robot control mechanisms (waypoints, speed, remote control) and publish the robot localisation information and odometry. It shall:

- Process and execute robot steering commands (either waypoints or real-time remote control commands).
- Control the speed of the robot and enforce any safety rule given: no-go areas, minimal or maximal altitude or depth, collision avoidance.
- Estimate and publish the robot odometry and any other localisation and speed information
- Monitor the vehicle critical resources such as the battery. Take safety measures (e.g. return to base) if energy is too low to complete the mission.
- Publish identification information.

Id:	UXV-NOD-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	Consortium	Ver:	2
Title:         Each UxV shall have a unique Identification code.									
Desc	<b>Description:</b> Each UxV shall have a unique Identification code across the testbed								
Addi (con	Additional Info (comments):This allows each system to be unequivocally identified in the RAWFIE network (comments):Messages transmitted across the network can be addressed to identify recipulation by using this unique identification code.						l netwo v recipi	ork. ient	
Com Subs	ponent or system								
Refi	nes/Replaces	TB-R-0	03						

Id:	UXV-NOD-002	Туре:	FUNC	Importance (priority):	HIGH	Source:	DoW	Ver:	2
Title:		Each U	xV node sho	uld ensure a m	inimum auto	nomy of 1	5-30 minutes.		
Description:	Multiple UxVs will provide to the experimenters a minimum duration of 45 to 90 minutes per session.								
--------------------------------	---								
Additional Info (comments):	Several current UxV platforms are capable of providing more autonomy, including the ones already available to the RAWFIE consortium. This figure is conservative to expand the range of UxV systems that can be added to the RAWFIE network (e.g: aerial vehicles) while still providing a minimum amount of autonomy to ensure functionality to the testbed.								
Component or									
Subsystem									
Refines/Replaces	TB-R-007								

Id:	UXV-NOD-003	Туре:	FUNC	Importance (priority):	HIGH	Source:	DoW	Ver:	2
Title	:	Each U	ach UxV node should ensure payload.						
Desc	cription:	Multiple UxVs will provide to the experimenters a minimum payload of 0.5-1k per unit.						1kg	
Addi (com	tional Info ments):	Several the on conserv RAWF of paylo	Several current UxV platforms are capable of providing more payload, including the ones already available to the RAWFIE consortium. This figure is conservative to expand the range of UxV systems that can be added to the RAWFIE network (e.g: aerial vehicles) while still providing a minimum amount of payload to ensure functionality to the testbed.						
Com Subs	ponent or ystem								
Refi	nes/Replaces	TB-R-0	08						

## 4.3.3 UxV Network and Communication

Id:	UXV-NET-001	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	2	
Title:		Capability of taking the control of the UxVs from distance.								

٦

Description:	The UxV shall support the possibility to be remotely controlled. It shall include a communication system and a control system that allow for its control remotely.
Additional Info (comments):	The UxVs will typically fly under a local control loop, heading to a waypoint, while being monitored by the RAWFIE system. In some circumstances, the UxV may need assistance (for precise action, landing, crossing a river) which can be provided by remote control. This implies the provision the appropriate communication quality of service, such as real-time guarantees. The corresponding technical requirements must be specified on a case by case basis, since they depend on the type of UxV, its mission and environment.
Component or Subsystem	UxV Network and Communication
Refines/Replaces	TB-R-006, TB-R-013

Id:	UXV-NET-002	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	2	
Title	:	UxVs s	hould be able	e to Synchroni	ze their Time	e-Reference	ces between the	em.		
Desc	cription:	The UxV shall include a mechanism for adjusting its local time reference of regular basis or on demand, with respect to an external time reference. objective is to share the same time reference (within a specific error ran across all UxVs in a given set (e.g. test-bed, experiment, swarm)							n a Γhe ge)	
Addi (com	tional Info ments):	The UxVs will typically use its local clock, which will drift over time. The Uz shall include a clock synchronization mechanism relying on the communicati system. The error range will depend on the chosen mechanism, the refresh rate and t quality of the local clocks, in addition to external factors, such as temperature.							xV ion the	
Com Subs	ponent or ystem	UxV No	UxV Network and Communication							
Refi	nes/Replaces	TB-R-0	TB-R-011							

1



Id:	UXV-NET-003	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	2
Title	:	The Ux	he UxV should provide Access Point functionality.						
Desc	ription:	The UxV shall embed a local access point feature.							
Addi (com	tional Info ments):	To be c request	lefined (Wha ). This requir	at is the functi rement was ide	onal need ar ntified based	nd the use l on define	case behind the discension case behind the discension of the discension of the discension of the discrete set of the discrete	he feat	ure
Com Subs	ponent or ystem	UxV Network and Communication							
Refi	nes/Replaces	TB-R-0	12						

Id:	UXV-NET-004	Туре:	FUNC	Importance (priority):	HIGH	Source:	Consortium	Ver:	2
Title	:	Each U means.	xV node sha	all be equipped	d with prima	ary and se	condary comm	nunicat	ion
Desc	ription:	The UxV shall include at least two communication systems.							
Addi (com	tional Info ments):	To be c request of one c	To be defined (What is the functional need and the use case behind the feature request): This can be used also for redundancy, which would allow for the failure of one communication system.						
Com Subs	ponent or ystem	UxV Network and Communication							
Refi	nes/Replaces	TB-R-0	TB-R-013						

Id:	UXV-NET-005	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	1	
Title:		UxV network interface management								

Description:	The UxV shall be able to detect, configure, control and use the network interfaces installed on the UxV specifically for communicate with the RAWFIE components.
Additional Info (comments):	
Component or Subsystem	UxV Network and Communication
Refines/Replaces	

Id:	UXV-NET-006	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	1
Title	:	UxV co	JxV communication interoperability with RAWFIE (incoming)						
<b>Description:</b> The UXV shall be able to receive and de-capsulate incoming messa. Testbed, deliver them to the relevant on-board component.					ming messages at.	s from	the		
Addi (com	tional Info ments):								
Com Subs	ponent or ystem	UxV Network and Communication							
Refi	nes/Replaces								

Id:	UXV-NET-007	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	1	
Title	2:	UxV co	JxV communication interoperability with RAWFIE (outgoing)							
Desc	cription:	The UX board c	V shall be omponents to	able to encaps o the RAWFIE	ulate and set platform via	nd messag a the Testl	ges originating bed.	from	on-	
Addi	itional Info									



(comments):	
Component or Subsystem	UxV Network and Communication
Refines/Replaces	

Id:	UXV-NET-008	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	1		
Title	:	Neighb	leighbouring UxV monitoring								
<b>Description:</b> The UxV shall be able to detect the presence and estimate the distance we neighbouring UxVs.							e with	the			
Addi (com	tional Info ments):	A local presence case the	e and the ese RAWFIE c	n shall be emb stimation of the ommunication	edded into t e distance w exhibits exc	the UxV ith its nei essive late	for the detecti ghbouring Ux encies.	on of Vs, in	the the		
Com Subs	ponent or ystem	UxV N	UxV Network and Communication								
Refi	nes/Replaces										

Id:	UXV-NET-009	Туре:	FUNC	Importance (priority):	HIGH	Source:	Consortium	Ver:	1	
Title	Title:Each UxV node should be able to send navigation state feedback with at lease Hz frequency and maximum 1 sec latency when within radio communicat reach.									
<b>Description:</b> Current radio communication technologies allow exchange of inf data in a network with high bandwidth and low latencies. We through a radio communication protocol, each UxV node must put two messages per second with state navigation information to Network latency shall be less than 1 sec.							unge of inform encies. When de must publis mation to the	ation a reacha sh at le netwo	and ble east ork.	
Addi (com	itional Info ments):	These s RAWF case sce	These should be considered as minimum requirements for a UxV to be used in RAWFIE experiments. Depending on the type of UxV and on a per application case scenario these constraints may become even stricter.							
Com Subs	ponent or system	UxV Network								



Refi	nes/Replaces			

## 4.3.4 UxV Sensor and Localisation

Id:	UXV-SEN-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title	:	Each U	xV node sho	ould tag location	n and timing	capability	to each sensor	r readir	ngs
<b>Description:</b> Sensors should provide to RAWFIE system measurement points, namely s information together with a timestamp (location information)						ely sen	sor		
Addi (com	tional Info ments):								
Com Subs	ponent or ystem	UxV Se	ensor and Lo	calisation					
Refi	nes/Replaces	TB-G-0	005, TB-R-00	)9					

Id:	UXV-SEN-002	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2
Title	:	Each U	xV node sha	ll be able to lis	t the availabl	le sensors			
Desc	cription:	Apart from listing the available sensors UxVs will describe the available Se Control Interface commands. Additionally, this is list shall be accessible f the UxV Network communication component directory service.						ole Sen sible fr	sor
Addi	tional Info								
(con	nments):								
Com	ponent or	UvV Se	nsor and Lo	calication					
Subs	ystem			cansanon					
Refi	nes/Replaces	TB-R-0	09						



Id:	UXV-SEN-003	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	1
Title	2:	UxV lo	cation and se	ensor data shou	ild be made a	available t	o the experime	nter	
<b>Description:</b> The experimenter by using the visualization tool will be able to viresource location and sensor data						able to view the	he curr	ent	
Add	itional Info								
(con	nments):								
Com	ponent or	U-V C		aliantian View		-1			
Subs	system	UXV Se	insor and Lo	callsation, vis	ualization too	IC			
Refi	nes/Replaces								

Id:	UXV-SEN-004	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2			
Title	:	Locatio during t	Location sensors should be supported in each UxV unit and can be used remotely during testbed demonstrations.									
Desc	cription:	UxV lo the next For exa waypoin informa able to	cation senso desired loca imple, throu nts of UxV tion should specify or to	rs should enab ation for each u gh the experin <sup>7</sup> will be sen be converted a change the loc	le users reme init. nenter contro it to the E and sent to the cation of each	otely throu oller the u ngine Co ne Visuali n unit.	ugh interfaces updated locatic ntroller. Then zation and use	to spec ons and locat rs will	l/or ion be			
Addi (com	itional Info nments):											
Com Subs	ponent or system	UxV Se	ensor and Lo	calisation – Ex	perimenter C	Controller	– Visualizatio	n tool				
Refi	nes/Replaces	TB-G-0	08									

Id:	UXV-SEN-005	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2

Title:	UxVs should sent a notification to the Resource Controller when they reach the desired location
Description:	The Resource Controller should be informed by the UxVs when all units reach the desired location. Additionally, apart from the current location they can share also information with regard their orientation and battery level
Additional Info	
(comments):	
Component or	
Subsystem	UxV Sensor and Localisation & Resource Controller
Refines/Replaces	

## 4.3.5 UxV On-board storage

Id:	UXV-STO-001	Туре:	DATA	Importance (priority):	HIGH	Source:	Consortium	Ver:	2			
Title		UxVs shall be able to store data on board.										
Desc	cription:	Capability of data storage, not only in case of transmission failure (Link loss between UxV's and the platform), but also upon user request. The data storage will also be needed for large data files because of the limited bandwith.										
Addi (com	itional Info ments):	The Ux exact po need to • Internal by the p	Vs usually operations back store data or Transmission retransmiss Sensors dar collected ar analysis aft data that me	communicate to c to the RAWF n board. Examp on failure (Lin sion of data as a ta content too nd stored and o ter the mission ay not be inter still can be info	heir sensor TE framewor ble cases incl ak loss betwo soon as the li large to be t lownloaded or experiment esting for the prmative for	measurem rk. In cert ude: een UxV' nk is estal ransmitted on the RA nt. e user (i.e manufactu	ents together ain cases there s and the plath blished again d in real time, WFIE platforn if it can't be runers etc)	with the will be form) a it will n for p ecogniz	ieir e a and be ost			
Com Subs	ponent or system	UxV O	UxV On-board storage									
Refi	nes/Replaces	TB-R-0	ГВ-R-004									



Id:	UXV-STO-002	Туре:	FUNC	Importance (priority):	HIGH	Source:	Consortium	Ver:	2		
Title	:	UxV's shall provide a management tool of the available storage.									
<b>Description:</b> Each UxV will need some tools not only for exchanging informati used or available storage, but also to offer a way of retrieving such deleting it in order to free the storage. The general functionality stollowing features:         • Permit to define a default configuration.         • Advertise available capacity.         Independently manage each data type						information so wing such shar ctionality shou	uch as ed data ld inclu	the a or ude			
Addi (com	tional Info ments):	Informa UxV as shall be	ation of the s a common used.	available stora message. Sha	ge can be en ared folders	xchanged and savin	within the sta g/deleting data	tus of a servi	the ces		
Com Subs	ponent or ystem	UxV C	UxV On-board storage								
Refi	nes/Replaces	TB-R-004									

Id:	UXV-STO-003	Туре:	SEC	Importance (priority):	HIGH	Source:	Consortium	Ver:	2		
Title	:	UxV's shall provide an authorized access to the data management tool.									
Desc	cription:	Access to data management tool need to be restricted only to authoriz personnel in order to avoid accidental overrides or deletions of data storage. A restricted retrieval of some data should be addressed too, given the spec nature of some kind of data (images etc.)									
Addi (com	tional Info ments):	Any kin Differe who an	nd of perman nt layers of d when can a	ent loss of the permissions g access the infor	data will nee iven to diffe mation.	ed to be au erent users	thorized and c	onfirm to spec	ed. xify		
Component or Subsystem											
Refi	nes/Replaces										

Id:	UXV-STO-004	Туре:	FUNC	Importance (priority):	HIGH	Source:	Consortium	Ver:	2	
Title	:	UxV's shall provide a data log.								
Desc	cription:	Any change in the data storage component shall be recorded in a log which we be available for querying at any time.								
Addi (com	tional Info ments):	The condata add	mponent wil dition/deletion	Il allow operate on or the retriev	or to gain av val of any kin	vareness ond of data.	of any change,	includ	ing	
Com	ponent or									
Subs	ystem									
Refi	nes/Replaces									

Id:	UXV-STO-005	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	2	
Title	:	UxV's may provide an automated syncing of servers.								
Desc	cription:	Semi-au periodio	utomated ma c upload of th	nagement of the data to a ren	he data may note server is	be provid s desirable	ed by the com	ponent	. A	
Addi (com	tional Info ments):	Particip should	bation of the be avoided.	operator as a	requirement	for the d	ata storage ma	nagem	ent	
Com Subs	ponent or ystem	UxV C	n-board stor	age						
Refi	nes/Replaces									

## 4.3.6 UxV On-board processing

The on-board processing aims at connecting data streams to on-board processing algorithms and publish the resulting output after checking for sufficient computing and energy resources. Allow the installation of new data processing algorithm and keep a registry.

Id:	UXV-PRC-001	Туре:	FUNC	Importance (priority):	HIGH	Source:	Consortium	Ver:	2		
Title	:	Each UxV shall be able to operate autonomously.									
Desc	cription:	The UxV shall be able to operate autonomously (without any external control) The objective is to give it the capability to make the flight as planned even in there are some disturbances, deviations, unexpected events, etc.									
Add (con	itional Info nments):	See also	D TB-REC-0	01							
Com Subs	ponent or system										
Refi	nes/Replaces	TB-R-0	01								

Id:	UXV-PRC-002	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	2		
Title	<u>.</u>	The UxV should provide collision avoidance mechanism.									
Description:		The UxV shall be able to autonomously avoid collision, for example by defining an "intimacy zone" in which no other object or UxV is allowed to enter without any specific reaction. However, reactive collision avoidance techniques shall reflect UxV type and environmental contraints. For example, while an AUV may stop its propeller to reduce momentum, and UAV may change height to avoid collision.									
		Finally, develop noise c collisio plannin	, since sens oment (e.g: : on acoustic ns do not o g stages.	ors tipically a reducing form echo sounder occur, and thu	vailable for -factor and s, etc.) the s, safety pro	collision price on ir perform ocedures	avoidance an laser systems, nance does n shall be appli	re still reduc ot ens ed at	in ing ure the		
Addi (con	itional Info ments):	See also	ee also TB-REC-001								
Com Subs	ponent or system										
Refi	nes/Replaces	TB-R-0	002								

-

Id:	UXV-PRC-003	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	2	
Title		Capability of task planning of the UxVs nodes during run-time.								
Desc	cription:	The user must have the capability to plan the course of a UxV and the tasks that it would have to execute during this course.								
Add (con	itional Info nments):	This red	quirement in	plies an appro	priate and ea	sy-to-use	User Interface			
Com	ponent or									
Subs	system									
Refi	nes/Replaces	TB-R-0	05							

-

Id:	UXV-PRC-004	Туре:	FUNC	Importance (priority):	MEDIUM	Source:	Consortium	Ver:	2
Title	:	UxVs s	UxVs should be able to cooperate during the execution of an experiment.						
Desc	ription:	The UxV should be able to exchange some data in real-time, at least with the nearest neighbour. This information may be used for the local and fit coordination inside or between UxV swarms or for cooperative monitoring of area.							the ine an
Addi (com	tional Info ments):	This rec	quirement im	plies an approp	priate and ea	sy-to-use	User Interface		
Com Subs	ponent or vstem								
Refin	nes/Replaces	TB-R-0	10						
	1	1 D-K-0	10						

Id:	UXV-PRC-005	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2		
Title:		Each U	Each UxV node shall keep position while waiting for new instructions.								

Description:	Each UxV node must keep its position (either stopped in a location, or moving within a contained radius) while waiting for new instructions from the RAWFIE software toolchain.
Additional Info (comments):	While system like UGVs usually stay at a fixed position when not actuated, other systems like USVs or UAVs are unable to keep position (not the case for rotary wing UAVs). This happens for UxV nodes that are underactuated. To ensure safety and guarantee nodes are kept within network reach, each UxV shall have a built-in software routine that prevents drifting from a defined region.
Component or	
Subsystem	
Refines/Replaces	

## 4.3.7 UxV Management

The UxV management provides a centralised dashboard view and control of the UxV operations and resources. It keeps a searchable registry of the UxV functions and resources.

Id:	UXV-MGT-001	Туре:	ОТН	Importance (priority):	HIGH	Source:	Consortium	Ver:	2	
Title	:	UxVs shall offer on demand resources (Network, Sensor, Processing, and Controller).								
Desc	cription:	Resources as Network connectivity, sensor readings and low level controller shall be offered within the Rawfie platform, taking into account safety an functional conditions.							ers	
Addi (com	tional Info ments):	The int sharing	egrity of the these resour	UxV's operat	ility shall no	ot be put i	n danger by ar	ıy kind	l of	
Com Subs	ponent or ystem	UxV M	UxV Management							
Refi	nes/Replaces	TB-NF	TB-NF-R-001							

Id:	UXV-MGT-002	Туре:	SEC	Importance (priority):	HIGH	Source:	Consortium	Ver:	2
Title:		UxV sh	all be capabl	e to revert to a	safe mode				

Description:	If needed, the UxV shall be capable of aborting any harmful process for the rawfie platform or itself, and revert to a safe mode. In this mode, functionality is limited and external supervision is required before returning to normal operation.
Additional Info (comments):	Operation errors forcing the safe mode should be specified, as well as the actuation protocol in these cases.
Component or Subsystem	UxV Management
Refines/Replaces	TB-NF-R-003

Id:	UXV-MGT-003	Туре:	FUNC	Importance (priority):	HIGH	Source:	Consortium	Ver:	2	
Title		UxV shall be capable to restart its internal components independently								
Desc	cription:	A malfu UxV's rest of t	unction in an shall provid he system	y UxV interna e a method of	l component performing	may need such resta	l an external re art without aff	start of	f it. the	
Addi (com	itional Info nments):									
Com Subs	ponent or system	UxV N	lanagement							
Refi	nes/Replaces									

Id:	UXV-MGT-004	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2		
Title:		UxV shall be capable to monitor the health of its components and provide appropriate health status messages to the testbed									
Description:		UxV sh situation	all monitor and the second sec	not only if need ors can be adve	ded compone ertised to the	ents are ru testbed.	nning smoothl	y but a	ılso		



Additional Info (comments):	Schemas and message formatting have already been discussed for this purpose. UxV should make use of them.
Component or Subsystem	UxV Management
Refines/Replaces	

Id:	UXV-MGT-005	Туре:	FUNC	Importance (priority):	HIGH	Source:	Iteration1 Exp	Ver:	2		
Title	:	UxV shall be capable to enable/disable certain internal components									
Desc	cription:	If need indeper The dis queried	If needed, the UxV shall be able to disable some of its internal component independently provided that these are not essential for the UxV function. The disablement of components may be forced due to new operation errors or queried by the user in order to comply with regulations or save energy.								
Addi (com	tional Info ments):	Should be remarked that enable/disable operation would require supervision of authorized personnel.									
Com Subs	ponent or ystem	UxV Management									
Refi	nes/Replaces										

Id:	UXV-MGT-006	Туре:	ОТН	Importance (priority):	HIGH	Source:	Consortium	Ver:	2		
Title	2:	UxV shall be capable to offer safe maintenance access for manufacturers									
Desc	cription:	In orde anytime conditio	r to reprogr e, according on.	am or update to the operatin	the system, g schedule o	manufact f the robo	turers shall ha t or given an e	ve acc mergei	ess ncy		
Addi (con	itional Info ments):										
Com	ponent or	UxV N	lanagement								



Subsystem	
Refines/Replaces	

## 4.4 Ethics and Security Requirements

In the first version of the Requirements' deliverable, a number of mainly non-functional requirements were defined both at Platform and at Testbed level with the purpose to act as a starting point for defining an architecture as well as methods and procedures that will provide a great degree of shielding against external spurious or malevolent actions. These requirements are presented in Table 5. They address to some extend the various ethics issues listed below that were raised in the RAWFIE DoW:

- 1. Dual use "Details on potential dual use implications of the project and risk-mitigation strategies must be provided and copy of ethics approval must be forwarded to Commission (if applicable)."
- 2. Misuse: Details on measures to prevent malevolent/criminal/terrorist abuse of research findings must be provided.
- 3. Misuse: "Ensure an enhanced, highly encrypted security protocol, that protect mobile units against hacking, being reprogrammed, and potentially used them for malevolent/criminal/terrorist abuses."
- 4. Misuse: "Ensure and integrate a non re-programmable and non modifiable read-only) code session within all mobile units that automatically send information to mobile-unit owner if a mobile-unit is remotely reprogrammed and allow mobile-unit owner to be able remotely immediately switch the unit off (with non re-programmable and non modifiable, read-only code session) if the change was not initiated by the mobile-unit owner."

ID	Category	Title	Туре	Priority	Source
		RAWFIE platform shall support secure			
PT-NF-001		data exchange	SEC	HIGH	DoW
		RAWFIE platform shall provide a			
		reservation/booking system with			
PT-NF-002		adequate security and privacy	SEC	HIGH	Consortium
		RAWFIE platform should be able to			
PT-NF-003		support backups of all critical data	SUPP	<b>MEDIUM</b>	DoW
		The Testbed infrastructure should			
		provide reliability and robustness of all			
TB-NF-G-002	General	components/modules.	SUPP	<b>MEDIUM</b>	Consortium
		The communication interfaces shall			
TB-NF-G-004	General	offer security mechanisms	SEC	HIGH	Consortium



		UxV shall be capable to revert to a safe						
TB-NF-R-003	Resource	mode	SEC	HIGH	Consortium			
Table 5: Iteration 1 Requirements that remain valid and relate to Ethical issues								

Table 5: Iteration 1 Requirements that remain valid and relate to Ethical issues

These are also certain functional requirements from iteration 2 that adhere to security and privacy issues

	Componen				
ID	t	Title	Туре	Priority	Source
	Experiment	RAWFIE shall provide a validator			
	Validation	to constantly check experiment			
PT-EXV-S-001	Service	scenarios during runtime	FUNC	HIGH	DoW
		Each Testbed should provide the			
		exact boundaries within which its			
TB-GEN-R-002	General	UxVs can operate	ENV	HIGH	Other
		RAWFIE platform should be able			
	Resource	to activate the "Emergency		MEDIU	Iteration1
<b>TB-REC-002</b>	Controller	Scenario"	FUNC	Μ	Exp
	UxV				
	Manageme	UxV shall be capable to revert to			Consortiu
UXV-MGT-002	nt	a safe mode	SEC	HIGH	m

Table 6: Iteration 2 Requirements that relate to Ethical issues

Specification & Analysis of RAWFIE Components Requirements (b)

# 5 Overview table and Traceability Mapping

In this section we provide a traceability matrix which can be used to track D3.2 requirements to D3.1 requirements.

No	חו	Component	Catagory	Titlo	Typo	Driority	Sourco	Vorsion	Iteration 1	Iteration 1
1		Conoral		PAWEIE Distform should adopt Sliced Endersted Architecture (SEA)	БЦМС	PHONE	Itoration1 Eve	version		
Ł	FI-GEN-R-001	General	FLATIORM	RAWFIE platform shall support various roles with different privileges	TONC	nion		2	F1-F-001	F 1-INI-008
2	PT-GEN-R-002	General	PLATFORM	at every level of access.	FUNC	HIGH	DoW	2	PT-GEN-002	
				The RAWFIE Data model should include all basic entities that are used						
3	PT-GEN-R-003	General	PLATFORM	or/and exchanged by the various components of the RAWFIE Platform	DATA	HIGH	Architecture Deliverables	2	PT-P-005	
				RAWFIE platform shall provide appropriate data storage for						
1	PT-GEN-R-004	General		information that needs to be persisted, or used after an experiment	ΓΛΤΛ	нсн	Iteration1 Evn	2	PT_P_005	
	TT GEN IX 004	General		A web portal interface shall be provided to the users of the platform	DAIA	mon		۷.		
5	PT-WEB-P-001	Web Portal Tool	PLATFORM	to access almost all main functionalities.	FUNC	HIGH	DoW	2	PT-GEN-001	
6	PT-WEB-P-002	Web Portal Tool	PLATFORM	Web portal usage shall be allowed only to authenticated users	FUNC	HIGH	DoW	2	PT-GEN-003	
				A tutorial or similar type of documentation shall be provided to the						
7	PT-WEB-P-003	Web Portal Tool	PLATFORM	users of the platform	FUNC	HIGH	DoW	2	PT-P-002	
				Booking Tool should allow booking of resources at the experimenter						
8	PT-BOO-T-001	Booking Tool	PLATFORM	level for a specified period and for selected resources	FUNC	HIGH	Iteration1 Exp	2	PT-B-001	
				Booking Tool functionality shall be compatible with the SFA myslice				_		
9	PT-BOO-1-002	Booking Iool	PLATFORM	architecture and the notion of slices reservations	FUNC	HIGH	Iteration1 Exp	2	PI-B-001	
10		Deeking Teel		Booking I ool should delegate all its actions related to Booking of a resource to the Booking Service	FUNC		Architactura Dalivarahlaa	2	DT D 001	
10	P1-BOO-1-003	BOOKING TOOI	PLATFORM	Tesource to the booking service	FUNC	HIGH	Architecture Deliverables	2	P1-B-001	
				Realing Tool shall also interact with the Testhads Directory Convice in						
11	РТ-ВОО-Т-004	Booking Tool	PLATFORM	order to retrieve information on unallocated testbed resources	FUNC	HIGH	Iteration1 Exp	2	PT-B-001	
				Booking Tool should communicate with the underline services using				_		
12	PT-BOO-T-005	Booking Tool	PLATFORM	JSON formatted messages (through an RPC or REST API)	FUNC	HIGH	Iteration1 Exp	2	PT-B-001	
				Booking Tool should provide appropriate functionality for viewing the						
13	PT-BOO-T-006	Booking Tool	PLATFORM	reservations of a user/experimenter	FUNC	HIGH	Architecture Deliverables	2	PT-B-002	
14	PT-BOO-T-007	Booking Tool	PLATFORM	Booking Tool should allow editing of existing Reservations	FUNC	HIGH	Architecture Deliverables	2	PT-B-002	
15	PT-BOO-T-008	Booking Tool	PLATFORM	Booking Tool should allow cancellation of existing Reservations	FUNC	HIGH	Architecture Deliverables	2	PT-B-002	
16		Booking Tool	ΡΙΔΤΕΩΒΜ	BOOKING TOOL Should allow creation of bookings through an intuitive UI	FUNC	HIGH	Architecture Deliverables	2	PT-B-002	
10				Appropriate notification mechanism should be provided to the user in	Tone	mon		~ ~	110002	
17	PT-BOO-T-010	Booking Tool	PLATFORM	case status of reservation request is not directly available.	FUNC	HIGH	Architecture Deliverables	2	PT-B-002	
				Booking Tool may provide assistance of feedback to the potential						
18	PT-BOO-T-011	Booking Tool	PLATFORM	experimenter during the booking process	FUNC	MEDIUM	Other	2	PT-B-005	
				Booking functionality should provide means to ensure fairness in						
10		Booking Tool		resource booking as well as protect for malevolent actions that a user	FUNC	нсч	Iteration1 Evn	2	PT_B_005	
19	11-000-1-012			RAWEIE platform should allow virtualization of available LIXVs		nich		2	1-0-003	
20	PT-BOO-T-013	Booking Tool	PLATFORM	resources during reservation process	FUNC	LOW	Consortium	2	PT-B-006	
21	PT-SYM-T-001	System	PLATFORM	Listing and/or visualisation of current system health status shall be	FUNC	HIGH	Iteration1 Exp	2	PT-NF-007	

	Monitoring Tool		available						
	System								
22 PT-SYM-T-002	Monitoring Tool	PLATFORM	The current system health status should be grouped thematically.	FUNC	MEDIUM	Iteration1 Exp	2	NEW	
	System		Filtering of the accessible component health statuses by user						
23 PT-SYM-T-003	Monitoring Tool	PLATFORM	roles/rights should be possible.	FUNC	MEDIUM	Iteration1 Exp	2	NEW	
	System						_		
24 PT-SYM-T-004	Monitoring Tool	PLATFORM	The health statuses webpage should be updated automatically.	DATA	MEDIUM	Iteration1 Exp	2	NEW	
	System		The health status information should include a severity indication and				_		
25 PT-SYM-T-005	Monitoring Tool	PLATFORM	possibly textual information with additional details.	FUNC	HIGH	Iteration1 Exp	2	NEW	
	Resource		The UI interface shall illustrate testbed and UxV information of the						
26 PT-REE-T-001	Explorer Tool	PLATFORM	RAWFIE federation that the experimenters should take advantage of	FUNC	HIGH	DoW	2	PT-P-001	PT-P-003
	Resource	DIATEODIA	Desistuation of testhods and UV/s may be possible via the W/sh Deutel	FUNC		Iteration 1 Fun	2		
27 PT-REE-1-002	Explorer Tool	PLATFORIVI	Registration of testbeds and UXVs may be possible via the web Portai	FUNC	LOW	Iteration1 Exp	2	PT-P-004	
	Resource	DI ATEODIA	Resource Explorer tool shall allow for fine-grained resources'	FUNC			2		
28 PI-REE-1-003	Explorer Tool	PLATFORM	searches	FUNC	MEDIUM	Consortium	2	PT-A-016	
	Resource		Link to the Reaking Tool should be provided	ELINC		Consortium	2		
23 PT-REE-1-004		FLATFORIVI		FUNC	WEDIOW	Consolition	2	P1-P-001	P1-P-005
	Experiment		Experiment Description Language (EDL) shall be used as a language	FUNC	шси	Itoration1 Eva	2		
50 PT-EXA-T-001	Authorning Tool	PLATFORIVI	The EDL shall allow the definition of all pecessary requirements for an	FUNC	nion		2	PT-A-001	
31 PT-FXA-T-002	Authoring Tool	PLATFORM	experiment	FUNC	нісн	Iteration1 Exp	2	PT-A-002	
31 11 2/071 002	Evnoriment		For each defined experiment specific metadata, i.e. name, version	10110			-	1177002	
32 PT-FXA-T-003			date and description shall be defined	FUNC	MEDILIM	Consortium	2	PT-A-002	
32 11 2/01 003	Experiment		An experimentar shall be able to provide initial conditions and/or	TONC		Consortium	<u> </u>	1177002	
33 PT-FXA-T-00/			configuration parameters for an experiment		MEDILIM	Consortium	2	PT-A-009	
35 11 2//(1004	Functional		An experimentar shall be able to manage/guide the evailable backed			Consortium	<u>ک</u>		
24 DT_EXA_T_005	Experiment Authoring Tool		An experimenter shall be able to manage/guide the available booked	ELINC	нісн	Scenario	2	PT_A_004	DT-A-005
34 FT-LAA-1-003	Authorning 1001	FLATIONN	An experimentar shall be able to define the type of information to be	TONC	mon	Scenario	2	FT-A-004	F1-A-005
25 DT-EXA-T-006	Experiment Authoring Tool		An experimenter shall be able to define the type of information to be	ELINC	нісн	Iteration1 Evn	2	PT_A_006	
55 PT-EAA-T-000	Authorning 1001	FLATFORIVI	An experimenter shall be able to define the type of metrics to be	FUNC	пісп		2	PT-A-000	
	Experiment		gathered and/or stored during an experiment and/or per UxV						
36 PT-EXA-T-007	Authoring Tool	PLATFORM	resource	FUNC	HIGH	Scenario	2	PT-A-007	
	Experiment		An experimenter shall be able to provide navigation or movement						
37 PT-EXA-T-008	Authoring Tool	PLATFORM	directives during experiment authoring	FUNC	HIGH	Scenario	2	PT-A-008	
	Experiment		An experimenter should be able to create groups of LIXV/s resources						
38 PT-EXA-T-009	Authoring Tool	PLATFORM	for which specific directives will apply.	FUNC	MEDIUM	Scenario	2	PT-A-010	
	Experiment		A textual editor shall be provided for the authoring of RAWFIE						
39 PT-EXA-T-010	Authoring Tool	PLATFORM	experiments	FUNC	HIGH	DoW	2	PT-A-011	
	Experiment		A visual/graphical editor shall be provided for the authoring of						
40 PT-EXA-T-011	Authoring Tool	PLATFORM	RAWFIE experiments	FUNC	HIGH	DoW	2	PT-A-012	
	Experiment		Platform shall allow saving, editing and/or deletion of an experiment						
41 PT-EXA-T-012	Authoring Tool	PLATFORM	defined via EDL	FUNC	HIGH	Other	2	PT-A-015	
	Experiment		The visual editor should allow the definition of movement and	FUNC			2		
42 PI-EXA-1-013	Authoring Iool	PLATFORM	During outboring of an owneriment colorition of recourses should be	FUNC	AIGA	Utner	2	PT-A-012	
	Experiment		limited only to the ones previously reserved from the user at the						
43 PT-EXA-T-014	Authoring Tool	PLATFORM	foreseen time of experiment	FUNC	HIGH	Iteration1 Exp	2	NEW	

44	PT-EXA-T-015	Experiment Authoring Tool	PLATFORM	Validation of EDL script should be possible prior to or during saving	FUNC	HIGH	Iteration1 Exp	2	PT-L-002	
45	PT-EXA-T-016	Experiment Authoring Tool	PLATFORM	An experimenter shall have the means to define actions or tasks that should run on a periodic or ad hoc basis during execution of an experiment	FUNC	MEDIUM	Scenario	2	PT-L-010	
46	PT-EXM-T-001	Experiment Monitoring Tool	PLATFORM	Experiment Monitoring Tool shall provide overview of experiments of a user	FUNC	HIGH	DoW	2	PT-L-004	
47	PT-FXM-T-002	Experiment Monitoring Tool	PLATFORM	Experiment Monitoring and Visualisation should be integrated	FUNC	MEDIUM	Iteration1 Exp	2	NFW	
18		Experiment Monitoring Tool		Cancellation of running experiments should be possible via Web	EUNC	MEDILIM	Iteration1 Exp			
40	PT-EXIVI-1-005	UxV Navigation	PLATFORIVI	This component will provide to the user the ability to remotely	FUNC			2		
49	PT-NAV-T-001		PLATFORM	navigate a squad of UxVs through a user friendly interface.	FUNC	HIGH	DoW	2	PT-L-008	
50	PT-NAV-T-002	Tool	PLATFORM	The tool should provided some validation of user's instructions	FUNC	HIGH	Iteration1 Exp	2	NEW	
51	PT-NAV-T-003	UxV Navigation Tool	PLATFORM	UxV Navigation Tool should be available for the navigation of all moving resources	FUNC	HIGH	DoW	2	PT-L-008	
52	PT-NAV-T-004	UxV Navigation Tool	PLATFORM	UxV Navigation Tool should be available to read from the database a detailed version of the map of the available areas	FUNC	HIGH	Iteration1 Exp	2	NEW	
53	PT-VIS-T-001	Visualisation Tool	PLATFORM	The Visualisation Tool shall allow the visualisation of information about the running experiments, in tabular/graphical form	FUNC	HIGH	Architecture Deliverables	2	NEW	
54	PT-VIS-T-002	Visualisation Tool	PLATFORM	A 3D visualization should be available for the tracking of all moving resources	FUNC	MEDIUM	DoW	2	PT-L-006	
55	PT-VIS-T-003	Visualisation Tool	PLATFORM	The Visualisation Tool may allow visualisation of video streams coming from the experiment, and experiment's camera control	FUNC	LOW	Architecture Deliverables	2	NEW	
56	PT-VIS-T-004	Visualisation Tool	PLATFORM	The Visualisation Tool shall provide access to information / features associated to each UxV device on the geographic map	FUNC	HIGH	Architecture Deliverables	2	NEW	
57	PT-VIS-T-005	Visualisation Tool	PLATFORM	The Visualisation Tool shall allow organization and manipulation of multiple geographic layers	FUNC	HIGH	Architecture Deliverables	2	NEW	
58	PT-VIS-T-006	Visualisation Tool	PLATFORM	Possibility of Adding/Removing/Updating graphical widgets should be provided	FUNC	MEDIUM	Architecture Deliverables	2	NEW	
59	PT-VIS-T-007		PLATFORM	Possibility to display both actual and expected UxVs' route and position should be provided	FUNC	HIGH	Architecture Deliverables	2	NEW	
60	PT-DAA-T-001	Data Analysis Tool	PLATFORM	Analysis tool will provide interface to data engine.	FUNC	MEDIUM	Iteration1 Exp	2	PT-E-003	PT-E-002
61	PT-DAA-T-002	Data Analysis Tool	PLATFORM	Analysis tool will provide access to past experiments	FUNC	LOW	Iteration1 Exp	2	PT-E-003	PT-E-001
62	PT-DAA-T-003	Data Analysis Tool	PLATFORM	Analysis tool will provide ability to query message bus streams	FUNC	MEDIUM	Iteration1 Exp	2	PT-E-004	
63	PT-DAA-T-004	Data Analysis Tool	PLATFORM	Analysis tool will provide interface to end running jobs	FUNC	MEDIUM	Iteration1 Exp	2	PT-E-003	PT-E-004
64	PT-DAA-T-005	Data Analysis Tool	PLATFORM	Analysis tool will provide a simple metric selection interface, a view of the result stream & the iob status tab			r.		PT-E-003	PT-E-002
		Testbeds								
65	PT-DIR-S-001	Directory Service	PLATFORM	The Testbed Directory Service shall provide access to information on all Testbeds registered in RAWFIE	FUNC	HIGH	Architecture Deliverables	2	PT-P-003	
66	PT-DIR-S-002	Testbeds Directory	PLATFORM	The Testbed Directory Service should provide access to information on all Testbeds registered in RAWFIE according to predefined filters	FUNC	MEDIUM	Architecture Deliverables	2	NEW	

	Service								
67 PT-DIR-S-003	Testbeds Directory Service	PLATFORM	The Testbed Directory Service shall provide access to information about available resources (UxVs) belonging to the testbeds registered in RAWFIE	FUNC	нідн	Architecture Deliverables	2	NEW	
68 PT-DIR-S-004	Testbeds Directory Service	PLATFORM	The Testbed Directory Service should provide access to information onavailable resources (UxVs) belonging to the testbeds registered in RAWFIE, and according to predefined filters	FUNC	MEDIUM	Architecture Deliverables	2	NEW	
69 PT-DIR-S-005	Testbeds Directory Service	PLATFORM	The Testbed Directory Service shoud provide the possibility to register new testbeds in the RAWFIE platform, as well as to unregister (delete) testbeds from the platform	FUNC	HIGH	Architecture Deliverables	2	NEW	
70 PT-DIR-S-006	Testbeds Directory Service	PLATFORM	Some basic query capabilities should be provided	FUNC	MEDIUM	Architecture Deliverables	2	PT-A-016	
71 PT-DIR-S-007	Testbeds Directory Service	PLATFORM	The Testbed Directory Service shall provide the possibility to register new resources belonging to a specific testbed in the RAWFIE platform, as well as to unregister (delete) resources	FUNC	нідн	Architecture Deliverables	2	NEW	
72 PT-CPV-001	EDL Compiler and Validator	PLATFORM	A tool for translating EDL into user directives shall be provided	FUNC	HIGH	DoW	2	PT-A-003	
73 PT-CPV-002	EDL Compiler and Validator	PLATFORM	An experimenter should have the opportunity to use a code generation engine	FUNC	HIGH	DoW	2	PT-A-003	
74 PT-CPV-003	EDL Compiler and Validator	PLATFORM	Experiments defined via EDL shall be validated after their authoring	FUNC	HIGH	DoW	2	PT-A-014	
75 PT-CPV-004	EDL Compiler and Validator	PLATFORM	The compiler and validator should communicate with the authoring tool in order to transfer error indications and hints for solving them	FUNC	HIGH	DoW	2	NEW	
76 <b>PT-EXV-S-001</b>	Experiment Validation Service	PLATFORM	RAWFIE shall provide a validator to constantly check experiment scenarios during runtime	FUNC	нідн	DoW	2	PT-L-001	
77 PT-EXV-S-002	Experiment Validation Service	PLATFORM	The validation service should perform syntactic checking	FUNC	HIGH	DoW	2	PT-L-001	
78 PT-EXV-S-003	Experiment Validation Service	PLATFORM	The validation service should perform semantic checking	FUNC	нідн	DoW	2	PT-L-001	
79 PT-USR-S-001	Users & Rights Service	PLATFORM	User login credentials checking shall be provided	FUNC	HIGH	DoW	2	PT-GEN-002	
80 PT-USR-S-002	Users & Rights Service	PLATFORM	RAWFIE platform shall support various roles with different privileges at every level of access.	FUNC	HIGH	DoW	2	PT-GEN-002	
81 PT-USR-S-003	Users & Rights Service	PLATFORM	The Users & Rights Service may provide a proxy service for web application that do not check access rights.	FUNC	HIGH	Iteration1 Exp	2	NEW	
82 PT-BOO-S-001	Booking Service	PLATFORM	Booking Service shall support reservations of resources at both user level and experiment level	FUNC	HIGH	Iteration1 Exp	2	PT-B-001	
83 PT-BOO-S-002	Booking Service	PLATFORM	User level booking shall be triggered by the Booking Tool via a REST API.	FUNC	HIGH	Iteration1 Exp	2	PT-B-001	
84 PT-BOO-S-003	Booking Service	PLATFORM	Experiment level booking shall be triggered by the experimenter before issuing a manual or schedule launching of a validated experiment	FUNC	нідн	Iteration1 Exp	2	PT-B-001	PT-L-002

130/140

85	PT-BOO-S-004	Booking Service	PLATFORM	Experiment level booking shall support both immediate booking as well as booking at a future time	FUNC	нідн	Iteration1 Exp	2	PT-B-001
				Booking Service shall provide all the necessary methods to manage					
				the bookings including addition, modification and	FUNC			2	
86	PT-BOO-S-005	Booking Service	PLATFORM	cancellation/deletion operations	FUNC	HIGH	Architecture Deliverables	2	NEVV
07		De altra Casta	DIATEODIA	Booking Service shall be able to compute and return feedback on	FUNC			2	
87	P1-BOO-S-006	BOOKINg Service	PLATFORM	conflicting bookings for a provided booking request	FUNC	HIGH	Architecture Deliverables	2	NEVV
00		Dooking Convice		Reservation Data should be persistent in order to survive service	FUNC	шен	Iteration1 Eve	2	
00	PT-BOO-S-007	BOOKINg Service	PLATFORIVI	Historical data retrieval for Bookings/Reservations should be available	FUNC	поп		Ζ	
89	PT-BOO-S-008	Booking Service	PLATFORM	on demand	FUNC	MEDIUM	Iteration1 Exp	2	
		0		Booking functionality shall support reservation of resources involving					
90	PT-BOO-S-009	<b>Booking Service</b>	PLATFORM	multiple testbeds	FUNC	HIGH	Architecture Deliverables	2	PT-B-003
				Booking functionality shall be able to correctly handle simultaneous					
91	PT-BOO-S-010	Booking Service	PLATFORM	Reservations requests by end users	FUNC	HIGH	Iteration1 Exp	2	РТ-В-003
				Notification mechanisms may be provided for experiments scheduled					
92	PT-BOO-S-011	Booking Service	PLATFORM	for execution in the future.	FUNC	MEDIUM	Consortium	2	PT-B-004
		Launching		Launching Service shall support short-term or manual launching of an					
93	PT-LAU-S-001	Service	PLATFORM	experiment initiated directly by an experimenter	FUNC	HIGH	Architecture Deliverables	2	NEW
		Launching		Launching Service shall support long-term or scheduled launching of					
94	PT-LAU-S-002	Service	PLATFORM	an experiment initiated directly by an experimenter	FUNC	HIGH	Architecture Deliverables	2	NEW
05		Launching		Each executing experiment shall be uniquely identified within RAWFIE	FUNC	шен	Architactura Dalivarablac	2	
95	PT-LAU-5-003	Service	PLATFORIVI		FUNC	нісн	Architecture Deliverables	Z	PT-E-001
06		Launching		During launching it must be ensured that the experiment to be started	FUNC	нісн	Architactura Dalivarablac	2	
90	PT-LAU-3-004		PLATFORIVI		FUNC	поп	Architecture Deliverables	ζ	PT-L-002
97		Launching		During launching it must be ensured that the experiment to be started belongs to an authorized user of the RAWEIE platform	FUNC	нісн	Architecture Deliverables	2	PT-1-002
57	11 LAO 3 003			The Launching Service chall be able to address simultaneous requests	TONC	man	Architecture Denverables	2	
98	PT-LALL-S-006	Service	ΡΙΔΤΕΟΡΜ	for starting an experiment	FUNC	нісн	Architecture Deliverables	2	NEW
50	11 210 3 000			The Launching Service shall send an appropriate message upon	Tone		Areinteetare Denverables	<u> </u>	
99	PT-LAU-S-007	Service	PLATFORM	successful starting of an experiment	FUNC	нідн	Architecture Deliverables	2	PT-F-001
				The Launching Service shall interact with other components or					
		Launching		database services in order to retrieve information needed for deciding					
100	PT-LAU-S-008	Service	PLATFORM	on launching an experiment	FUNC	HIGH	Architecture Deliverables	2	NEW
		Launching		Interactions of the launching service with database services and/or					
101	PT-LAU-S-009	Service	PLATFORM	other components should respect the RAWFIE platform boundary	FUNC	HIGH	Iteration1 Exp	2	NEW
102		Launching			FUNC		Iteration 1 From	2	
102	PT-LAU-S-010	Service	PLATFORIVI	Launching service shall support requests for experiment cancellation	FUNC	HIGH	Iteration1 Exp	2	
103	PT-LAU-S-011	Service	PLATFORM	experiments execution	FUNC	MEDIUM	Consortium	2	PT-L-007
		Launching		Launching service shall provide appropriate feedback to the				_	
104	PT-LAU-S-012	Service	PLATFORM	requested entity regarding failures on fulfilling a request	FUNC	HIGH	Iteration1 Exp	2	NEW
		Launching		Launching service shall not alter or modify any information related to					
105	PT-LAU-S-013	Service	PLATFORM	the actual execution of an experiment	FUNC	HIGH	Iteration1 Exp	2	NEW
		Visualisation		The Visualization Engine shall retrieve from the message bus all					
106	PT-VIS-E-001	Engine	PLATFORM	runtime experiment information needed for visualizing the UxVs	FUNC	HIGH	Architecture Deliverables	2	PT-L-005

sis	of RAWFIE	Components	Requirements	(b)
				· ·

				and/or any sensor measurments						
107	PT-VIS-E-002	Visualisation Engine	PLATFORM	The Visualization Engine shall provide a GIS server capable of handling geographical layers (overlays)	FUNC	HIGH	Architecture Deliverables	2	NEW	
		Visualisation		The Visualization Engine may allow cache of data for faster access to						
108	PT-VIS-E-003	Engine	PLATFORM	the available geographic layers	FUNC	MEDIUM	Architecture Deliverables	2	NEW	
		Visualisation		The Visualization Engine shall provide the possibility to reply						
109	PT-VIS-E-004	Engine	PLATFORM	experiments using historical data	FUNC	HIGH	Architecture Deliverables	2	NEW	
		Experiment								
110	PT-EXP-C-001	Controller	PLATFORM	Cancellation of running experiments should be possible	FUNC	HIGH	Iteration1 Exp	2	NEW	
		Experiment	DIATEODIA	RAWFIE platform shall allow experimenters to remotely navigate	FUNC			2		
111	PT-EXP-C-002	Controller	PLATFORM	UXVS.	FUNC	MEDIUM	Consortium	2	PT-L-008	
		Experiment		The Experiment Controller shall support the execution of experiments						
112	PT-EXP-C-003	Controller	PLATFORM	that involve multiple testbeds	FUNC	HIGH	Iteration1 Exp	2	NEW	
		Experiment		The Experiment Controller shall be able to support multiple						
113	PT-EXP-C-004	Controller	PLATFORM	experiments running the same time in parallel	FUNC	HIGH	Iteration1 Exp	2	NEW	
		E		The Experiment Controller shall be able to analyse the whole						
114		Experiment		experiment script and dispatch the appropriate parts to each	ELING	шсц	Itoration1 Eva	n		
114	PT-EXP-C-005	Controller	PLATFORI	The Experiment Controller shall support receiving feedback at regular	FUNC	поп		2		
		Experiment		intervals from all testbed facilities about the progress of the						
115	PT-EXP-C-006	Controller	PLATFORM	experiment in this time interval	FUNC	нідн	Iteration1 Exp	2	NEW	
				The Experiment Controller may be able to override the order of						
		Experiment		instructions described in the input script while the experiment is						
116	PT-EXP-C-007	Controller	PLATFORM	running	FUNC	HIGH	Iteration1 Exp	2	NEW	
				The Experiment Controller shall be able to continuously feed the						
		Experiment		front-end tier (Experiment Monitoring Tool) giving the experimenter a						
117	PT-EXP-C-008	Controller	PLATFORM	clear view of the experiment workflow as a whole	FUNC	HIGH	Iteration1 Exp	2	PT-L-004	
				The Experiment Controller shall send distinct error and warning						
		Experiment		messages in every case the experiment's state diverges from the						
118	PT-EXP-C-009	Controller	PLATFORM	aimed target	FUNC	HIGH	Iteration1 Exp	2	NEW	
		Data Analysis								
119	PT-DAA-S-001	Engine Data Analysia	PLATFORM	Analysis engine will support accepting of analysis jobs	FUNC	MEDIUM	Iteration1 Exp	2	PT-E-004	PT-E-005
120		Engine		Analysis anging will support compiling analysis jobs	FUNC	MEDILIM	Iteration1 Evn	2	PT_F_005	
120	11-DAA-3-002	System			TONC			2	11-6-005	
		Monitoring		RAWFIE middle tier shall include a module to monitor the						
121	PT-SYM-S-001	Service	PLATFORM	performance of the middle tier components.	FUNC	HIGH	Consortium	2	PT-GEN-004	
		System								
		Monitoring								
122	PT-SYM-S-002	Service	PLATFORM	RAWFIE Testbeds and UxVs statuses should be monitored	FUNC	HIGH	Iteration1 Exp	2	NEW	
		System								
4.2.2		Monitoring	DIATEODIA	RAWFIE system administrators should be informed if critical, for the	FUNC			2		
123	PT-SYIVI-S-003	Service	PLATFORM	RAWFIE platfrom operation, services are down	FUNC	HIGH	Iteration1 Exp	2	PT-NF-007	
		Monitoring								
124	PT-SYM-S-004	Service	PLATFORM	User may register for notifications if certain components are down	FUNC	LOW	Iteration1 Fxn	2	PT-NF-007	
125	PT-SYM-S-005	System	PLATFORM	Notifications about planned downtimes	FUNC	MEDIUM	Iteration1 Exp	2	PT-NF-007	

		Monitoring Service								
126	PT-ACC-S-001	Accounting Service	PLATFORM	The accounting service should be capable to accept different cost models regarding RAWFIE usage on a per service basis	FUNC	MEDIUM	DoW	2	PT-B-007	
127	PT-ACC-S-002	Accounting Service	PLATFORM	The accounting service should be capable to gather statistics regarding usage of the platform by experimenters.	FUNC	MEDIUM	DoW	2	PT-B-007	
128	PT-ACC-S-003	Accounting Service	PLATFORM	The RAWFIE platform should record information related to time and type of access for a service by a user.	FUNC	MEDIUM	DoW	2	PT-B-007	
129	PT-ACC-S-004	Accounting Service	PLATFORM	The cost model used may take into consideration the overall time of experiments executed by a user of the platform.	FUNC	MEDIUM	Iteration1 Exp	2	РТ-В-007	
130	PT-ACC-S-005	Accounting Service	PLATFORM	The accounting service may support different types of charging based on the type of the experimenter (industrial, research, university etc.)	FUNC	MEDIUM	Iteration1 Exp	2	PT-B-007	
		Accounting		The accounting service may support predefined types of memberships regarding usage of the platform that may depend on various types of						
131	PT-ACC-S-006	Service	PLATFORM	parameters The accounting service should be able to bandle the addition of new	FUNC	MEDIUM	Iteration1 Exp	2	PT-B-007	
		Accounting		services that may be incorporated in the RAWFIE platform during						
132	PT-ACC-S-007	Service	PLATFORM	time.	FUNC	MEDIUM	Iteration1 Exp	2	PT-B-007	
				Each UxV Testbed should provide a Slice Interface for federating their						
133	TB-GEN-R-001	General	TESTBED	capabilities/resources to the experimenter.	FUNC	HIGH	Iteration1 Exp	2	NEW	
124		Conorol	TECTOED	Each Testbed should provide the exact boundaries within which its			Other	2		
134	IB-GEN-R-002	General	TESTBED	UXVS can operate Testhed areas should at least he able to host (operate multiple Liv)/s	EINV	HIGH	Other	2	INEVV	
135	TB-GEN-R-003	General	TESTBED	of one or more types	FUNC	нідн	Other	2	NFW	
136	TB-GEN-R-004	General	TESTBED	Testbed areas environment should be closely monitored	FNV	HIGH	Other	2	TB-G-002	
100				Indoor spaces of a testbed should provide a controlled indoor						
137	TB-GEN-R-005	General	TESTBED	environment	ENV	HIGH	Other	2	TB-G-002	
				Testebed facility areas should comprise storing spaces and be able to						
138	TB-GEN-R-006	General	TESTBED	receive inspect and assemble and/or fix UxVs	SUPP	HIGH	Other	2	TB-G-002	
				Testbed facilities should provide emergency services in an						
139	TB-GEN-R-007	General	TESTBED	extraordinary event	SEC	HIGH	Other	2	TB-G-002	
140	TB-GEN-R-008	General	TESTBED	Testbed areas should provide proper facilities and equipment	ENV	HIGH	Other	2	TB-G-002	
141	TB-GEN-R-009	General	TESTBED	Testbed must provide dedicated computational resources	ENV	HIGH	Other	2	NEW	
142	TB-GEN-R-010	General	TESTBED	Testbeds should be supported by on-site personnel	OTH	HIGH	Other	2	NEW	
143	TB-GEN-R-011	General	TESTBED	Testbeds should conform to all legal regulations and restrictions	SEC	HIGH	Other	2	TB-NF-G-005	
144	TB-MOM-001	Monitoring Manager	TESTBED	The Monitoring Manager component should be able to provide information about the capabilities of each resource node.	DATA	HIGH	Iteration1 Exp	2	TB-G-004	TB-G-006
145	TB-MOM-002	Monitoring Manager	TESTBED	The Monitoring Manager component should collect and report current status of testbed facilities	DATA	HIGH	Iteration1 Exp	2	TB-G-001	
146	TB-MOM-003	Monitoring Manager	TESTBED	The Monitoring Manager component should store periodically all testbed information	DATA	HIGH	Iteration1 Exp	2	TB-G-003	
147	TB-MOM-004	Monitoring Manager	TESTBED	Testbed monitoring manager should be able to transmit the current status to the System Monitoring Service.	FUNC	HIGH	Iteration1 Exp	2	TB-G-003	
		Network	TECTORO	The RAWFIE communication resources shall be managed to offer						
148	IB-NEC-001	Controller	TESTBED	seamless connectivity in the normal operations of the system.	FUNC	MEDIUM	Consortium	2	TB-G-008	PI-L-009
149	TB-NEC-002	Network	TESTBED	Provision of network communication resource	FUNC	MEDIUM	Consortium	2		

		Controller								
		Network								
150	TB-NEC-003	Controller	TESTBED	Alternative communication system	FUNC	MEDIUM	Consortium	2	TB-R-013	
1 - 1		Network	TECTORD	Name and a fithe communication system	FUNC		Concertium	2		
151	TB-NEC-004	Network	TESTBED	Management of the communication system	FUNC		Consortium	۷	TB-INF-G-006	
152	TB-NEC-005	Controller	TESTBED	Time constraint verification and notification	FUNC	MEDIUM	Consortium	2	NEW	
		Resource		RAWFIE platform shall support a semi-autonomously way of						
153	TB-REC-001	Controller	TESTBED	navigation of the UxVs	FUNC	HIGH	Consortium	2	PT-L-008	TB-G-007
154		Resource	TESTRED	PAN/EIE platform should be able to activate the "Emorgancy Scoparia"	ELING		Itoration1 Evn	2		
154	TB-REC-002	Descurse	TESTBED	The Decourse Controller shall receive location messages from the	FUNC			2	PT-L-009	18-0-008
155	TB-RFC-003	Controller	TESTBED	vehicles at regular intervals	FUNC	HIGH	Iteration1 Exp	2	TB-G-005	TB-G-003
100		Resource		The Resource Controller shall transmit the next location for the						10 0 000
156	TB-REC-004	Controller	TESTBED	current experiment to the vehicles	FUNC	HIGH	Iteration1 Exp	2	TB-G-008	
				The Resource Controller shall be able to plan the next location that						
453	TD D50 005	Resource	TECTOR	will be transmitted in the vehicle taking into account the locations of						
157	TB-REC-005	Controller	TESTBED	all UxVs that are active in that testbed	FUNC	HIGH	Iteration1 Exp	2	NEW	
159		Resource	TESTRED	For the experiment accomplishment the Resource Controller shall	ELINC	нісн	Iteration1 Exp	2		TR-G-005
159	TB-PRO-001	Testhed Proxy	TESTBED	Testhed proxy should act as a reverse proxy	FUNC	MEDILIM	Consortium	2	NFW/	18-0-005
160	TB-PRO-002	Testbed Proxy	TESTBED	Testbed proxy contains Inner and Outer Firewall	FUNC	MEDIUM	Iteration1 Exp	2	NEW	
		Testbed		Testbed Manager shall support permanent storage of all testbed				_		
161	TB-MAN-001	Manager	TESTBED	attributes and resources attributes that belong to testbed	FUNC	HIGH	Consortium	2	TB-D-001	
		Testbed		Testbed Manager shall provide information about the capabilities of						
162	TB-MAN-002	Manager	TESTBED	each resource node	FUNC	HIGH	Consortium	2	TB-G-004	
4.62	TO MANLOOD	Testbed	TECTORD	Testbed Manager shall check periodically the status of all other	FUNC			2		
163	TB-MAN-003		TESTBED	services running at testbed level	FUNC	HIGH	Iteration1 Exp	2	NEVV	
164	TB-MAN-004	lestbed Manager	TESTBED	lestbed Manager shall contain a registration log for all the	FUNC	нісн	Iteration1 Exp	2	TB-D-002	
104		Testhed		Testhed Manager shall be periodically informed about the status of all	TONC	mon		2	10 0 002	
165	TB-MAN-005	Manager	TESTBED	running experiments in the testbed	FUNC	HIGH	Iteration1 Exp	2	NEW	
		Testbed		Testbed Manager shall store configuration parameters for the UxVs in			- '			
166	TB-MAN-006	Manager	TESTBED	the relevant testbed	FUNC	MEDIUM	Iteration1 Exp	2	TB-G-004	
		Testbed		Testbed Manager shall implement a user interface to support the						
167	TB-MAN-007	Manager	TESTBED	interactions between testbed operators and machines	FUNC	HIGH	Iteration1 Exp	2	NEW	
		Teethed		Testbed Manager shall be capable to handle temporary interruption						
168	TB-MAN-008	Manager	TESTBED	failure	FUNC	HIGH		2	TB-D-001	
100		Testbed		Testbed Manager may provide statistical data/information about	10110					
169	TB-MAN-009	Manager	TESTBED	testbed operation	DATA	LOW	Consortium	2	TB-D-002	
170	TB-UVG-001	General	UxV	Complianceof UxV to RAWFIE specification and interfaces	FUNC	HIGH	Iteration1 Exp	2	NEW	
171	UXV-NOD-001	UxV Node	UxV	Each UxV shall have a unique Identification code.	FUNC	HIGH	Consortium	2	TB-R-003	
172				Each UxV node should ensure a minimum autonomy of 15-30	FUNC			2		
172				Fach LivV node should ensure navioad	FUNC	HIGH		2	TB-R-009	
174	UXV-NFT-001	UxV Network		Capability of taking the control of the LixVs from distance	FUNC	MEDILIM	Consortium	2	TB-R-006	
- / /	5							-		

		and								
		Communication								
		and		LixVs should be able to Synchronize their Time-References between						
175	UXV-NET-002	Communication	UxV	them.	FUNC	MEDIUM	Consortium	2	TB-R-011	
		UxV Network						_		
		and								
176	UXV-NET-003	Communication	UxV	The UxV should provide Access Point functionality.	FUNC	MEDIUM	Consortium	2	TB-R-012	
		UxV Network								
		and		Each UxV node shall be equipped with primary and secondary						
177	UXV-NET-004	Communication	UxV	communication means.	FUNC	HIGH	Consortium	2	TB-R-013	
		UxV Network								
		and								
178	UXV-NET-005	Communication	UxV	UxV network interface management	FUNC	MEDIUM	Consortium	2	NEW	
		UxV Network								
170		and			FUNC					
179	UXV-INET-006	Communication	UXV	UXV communication interoperability with RAWFIE (incoming)	FUNC	MEDIUM	Consortium	2	NEVV	
		UXV Network								
180		Communication		LIXV communication interonerability with RAW/FIF (outgoing)	FUNC	MEDILIM	Consortium	2		
100	0/11/1-007		0.0		TONC		Consolition	2		
		and								
181	UXV-NET-008	Communication	UxV	Neighbouring UxV monitoring	FUNC	MEDIUM	Consortium	2	NEW	
		UxV Network		Each UxV node should be able to send navigation state feedback with						
		and		at least 2 Hz frequency and maximum 1 sec latency when within radio						
182	UXV-NET-009	Communication	UxV	communication reach.	FUNC	HIGH	Consortium	2	NEW	
		UxV Sensor and		Each UxV node should tag location and timing capability to each						
183	UXV-SEN-001	Localisation	UxV	sensor readings	FUNC	HIGH	Iteration1 Exp	2	NEW	
		UxV Sensor and								
184	UXV-SEN-002	Localisation	UxV	Each UxV node shall be able to list the available sensors	FUNC	HIGH	Iteration1 Exp	2	NEW	
		UxV Sensor and		UxV location and sensor data should be made available to the						
185	UXV-SEN-003	Localisation	UxV	experimenter	FUNC	HIGH	Iteration1 Exp	2	NEW	
		UxV Sensor and		Location sensors should be supported in each UxV unit and can be						
186	UXV-SEN-004	Localisation	UxV	used remotely during testbed demonstrations.	FUNC	HIGH	Iteration1 Exp	2	NEW	
		UxV Sensor and		UxVs should sent a notification to the Resource Controller when they						
187	UXV-SEN-005	Localisation	UxV	reach the desired location	FUNC	HIGH	Iteration1 Exp	2	NEW	
		UxV On-board								
188	UXV-STO-001	storage	UxV	UxVs shall be able to store data on board.	DATA	HIGH	Consortium	2	TB-R-004	
		UxV On-board								
189	UXV-STO-002	storage	UxV	UxV's shall provide a management tool of the available storage.	FUNC	HIGH	Consortium	2	TB-R-004	
		UxV On-board		UxV's shall provide an authorized access to the data management						
190	UXV-STO-003	storage	UxV	tool.	SEC	HIGH	Consortium	2	NEW	
101		UXV Un-board		LIVI's shall provide a data las	FUNC	LUCU.	Concertions	2		
191	0.0.0.004	Storage	UXV	Oxy's shall provide a data log.	FUNC	HIGH	Consortium	2	INEVV	
102		storage		LIVV's may provide an automated syncing of convers	ELING		Consortium	2		
192	070-310-005	LIV/ On-board	0.00	ove s may provide an automated syncing of servers.	FUNC		Consol tium	2	INE VV	
193	UXV-PRC-001	processing	UxV	Fach UxV shall be able to operate autonomously	FUNC	нісн	Consortium	2	TB-R-001	
10/		Uv/ On board		The LIVV should provide collicion avoidance mechanism	ELINC		Consortium	2		
194	0/0-1/0-002		0.0	The one should provide consister avoluance mechanism.	TONC		Consolitium	2	10-11-002	

		processing								
195	UXV-PRC-003	UxV On-board processing	UxV	Capability of task planning of the UxVs nodes during run-time.	FUNC	MEDIUM	Consortium	2	TB-R-005	
196	UXV-PRC-004	UxV On-board processing	UxV	UxVs should be able to cooperate during the execution of an experiment.	FUNC	MEDIUM	Consortium	2	TB-R-010	
197	UXV-PRC-005	UxV On-board processing	UxV	Each UxV node shall keep position while waiting for new instructions.	FUNC	HIGH	Iteration1 Exp	2	NEW	
198	UXV-MGT-001	UxV Management	UxV	UxVs shall offer on demand resources (Network, Sensor, Processing, and Controller).	ОТН	HIGH	Consortium	2	TB-NF-R-001	
199	UXV-MGT-002	UxV Management	UxV	UxV shall be capable to revert to a safe mode	SEC	HIGH	Consortium	2	TB-NF-R-003	
200	UXV-MGT-003	UxV Management	UxV	UxV shall be capable to restart its internal components independently	FUNC	HIGH	Consortium	2	NEW	
201	UXV-MGT-004	UxV Management	UxV	UxV shall be capable to monitor the health of its components and provide appropriate health status messages to the testbed	FUNC	HIGH	Iteration1 Exp	2	NEW	
202	UXV-MGT-005	UxV Management	UxV	UxV shall be capable to enable/disable certain internal components	FUNC	HIGH	Iteration1 Exp	2	NEW	
203	UXV-MGT-006	UxV Management	UxV	UxV shall be capable to offer safe maintenance access for manufacturers	ОТН	HIGH	Consortium	2	NEW	

Table 7: Overview of Iteration 2 defined requirements including traceability to D3.1 Requirements

Based on the above traceability matrix the following table was created which includes requirements of D3.1 that do not have links to requirements defined in the present document. An extra column is provided for each such requirement specifying whether it is now OBSOLETE, already CONSIDERED or still VALID.

#		Iteration 1		
	Iteration 1 Reqs	Category	Iteration 1 Description	VALID
1	PT-A-013	Authoring Phase	Spatial information shall be provided for the currently available resources for the authoring of new experiments	
2	РТ-Е-002	Evaluation Phase	RAWFIE platform shall include a service enabling the data collection, analysis and processing.	YES
3	PT-L-003	Launching Phase	Launching tool shall be kept informed upon an experiment's state	OBSOLETE
4	PT-NF-001		RAWFIE platform shall support secure data exchange	YES
5	PT-NF-002		RAWFIE platform shall provide a reservation/booking system with adequate security and privacy	YES
6	PT-NF-003		RAWFIE platform should be able to support backups of all critical data	YES
7	PT-NF-004		RAWFIE platform shall exhibit high degree of network availability	YES
8	PT-NF-005		RAWFIE platform shall be able to support (near) real-time information gathering from the UxV sensors	YES
9	PT-NF-006		RAWFIE platform shall exhibit high degree of scalability	YES
10	PT-NF-009		RAWFIE architecture should adopt a modular design approach.	CONSIDERED
11	PT-NF-010		RAWFIE platform shall be deployed as a cloud based service (or list of services).	CONSIDERED
12	PT-NF-011		RAWFIE software modules should be implemented as Web Service or as REST	OBSOLETE
13	PT-NF-012		RAWFIE modules should use Open Standards and Open Software as far as possible	CONSIDERED
14	TB-G-009	General	The Testbed shall be able to support simulated UxVs resources	YES
15	TB-I-002	Interconnectivity	The communication system shall be able to use UxVs to relay information to and from other UxVs	OBSOLETE
	TB-I-003	Interconnectivity	A Testbed's communication system may provide at least 3 levels of Service and the communication means will adapt to	OBSOLETE
16			these Levels of Service	
17	TB-I-004	Interconnectivity	The Testbed shall be able to dispatch UxV information on demand	OBSOLETE
18	TB-NF-G-001	General	The Testbed shall provide concurrent requests capacity	CONSIDERED

### Specification & Analysis of RAWFIE Components Requirements (b)

19	TB-NF-G-002	General	The Testbed infrastructure should provide reliability and robustness of all components/modules.	YES
20	TB-NF-G-003	General	The communication system shall offer a high availability	YES
21	TB-NF-G-004	General	The communication interfaces shall offer security mechanisms	YES
22	TB-NF-R-001	Resource	UxVs shall offer on demand resources (Network, Sensor, Processing, and Controller).	YES
23	TB-NF-R-002	Resource	UxVs sensor system shall be compliant to connection standards and communication interfaces.	CONSIDERED
24	TB-R-009	Resource	Each UxV node should be equipped with a location identification system.	CONSIDERED

 Table 8: Not mapped Requirements of iteration 1 and their status regarding RAWFIE system

## Specification & Analysis of RAWFIE Components Requirements (b)



## 6 Conclusion

The present deliverable performs a more detailed requirements analysis for the RAWFIE platform, the testbed facilities and the UxVs to be used for experiments. Compared to the first version of the Requirements deliverable (D3.1) which focuses more in providing high level system requirements this one provides more fine grained requirements per component having as reference the components defined in the first version of the architecture (D4.1). The overall methodology and templates used are similar to the previous iteration with minor additions in order to support traceability between the requirements defined in each version of the requirement document.

While for iteration 1 requirements, we were based mainly in the DoA and the defined scenarios in the present document many requirements were defined based on feedback and experience gained from the first iteration design and development activities.



#### References 7

- [1] RAWFIE DOA AMENDMENT Reference No AMD-645220-14, Amendment AMD-645220-14.pdf, 7/12/2015
- [2] RAWFIE D 3 1 final.pdf RAWFIE Deliverable March 2015
- [3] D4.1 High Level Design and Specification of RAWFIE Architecture.pdf RAWFIE Deliverable June 2015
- [4] RAWFIE D4.2 645220 Design and Specification of RAWFIE Components (a).pdf RAWFIE Deliverable July 2015
- [5] International Organization for Standardization; ISO/IEC WD 29148.3; Software and Systems Engineering – Life Cycle Processes – Requirements Engineering, 2010.
- [6] Volere Template Edition 13, 2007, http://www.volere.co.uk/template.htm
- [7] Patibandla, S.T.; Bakker, T.; Klenke, R.H. "Initial evaluation of an IEEE 802.11s
- [8] S. Morgenthaler, T. Braun, Zhongliang Zhao, T. Staub and M. Anwander, "UAV Net: A mobile wireless mesh network using Unmanned Aerial Vehicles," Globecom Workshops (GC Wkshps), 2012 IEEE, pp. 1603
- [9] Thierry Rakotoarivelo, Max Ott, Guillaume Jourjon, Ivan Seskar, "OMF: a control and management framework for networking testbeds", in ACM SIGOPS Operating Systems Review 43 (4), 54-59, Jan. 2010.
- [10] http://www.ruby-lang.org
- [11] http://omf.mytestbed.net
- [12] Mathieu Lacage, Martin Ferrari, Mads Hansen, Thierry Turletti. NEPI: Using Independent Simulators, Emulators, and Testbeds for Easy Experimentation, ROADS 2009
- [13] Alina Quereilhac, Mathieu Lacage, Claudio Freire, Thierry Turletti and Walid Dabbous. NEPI: An integration framework for Network Experimentation, in proceedings of 19th International Conference on Software, Telecommunications and Computer Networks (SoftCOM), 2011
- [14] http://nepi.inria.fr
- [15] http://www.des-testbed.net/node/231
- [16] European Remotely-Piloted Aircraft Systems (RPAS) Steering Group (ERSG) Roadmap for the integration of civil Remotely-Piloted Aircraft Systems into the European Aviation System, Final report from the European RPAS Steering Group (June 2013) http://ec.europa.eu/enterprise/sectors/aerospace/files/rpas-roadmap\_en.pdf
- [17] European Commission, "A new era for aviation: Opening the aviation market to the civil use of remotely piloted aircraft systems in a safe and sustainable manner", Communication from the Commission to the European Parliament and the Council, http://eur-lex.europa.eu/legalcontent/EN/TXT/PDF/?uri=CELEX:52014DC0207&from=EN, COM(2014) 207 final



- [18] ParStream <u>https://www.parstream.com/</u>
- [19] RapidMiner <u>https://rapidminer.com/</u>
- [20] Apache Samoa <u>http://samoa.incubator.apache.org/</u>
- [21] Apache Storm <u>https://storm.apache.org/</u>
- [22] Apache Samza http://samza.apache.org/
- [23] Apache S4 <u>http://incubator.apache.org/s4/</u>
- [24] http://www.riot.ch/legal-information-about-flying-multicopter-drones-commercial/
- [25] <u>http://www.developpement-durable.gouv.fr/Quelle-place-pour-les-drones-dans.html</u> (in French)
- [26] https://okeanos.grnet.gr/home/